

Title	Control circuits for avalanche photodiodes		
Authors	Deng, Shijie		
Publication date	2013		
Original Citation	Deng, S. 2013. Control circuits for avalanche photodiodes. PhD Thesis, University College Cork.		
Type of publication	Doctoral thesis		
Rights	© 2013. Shijie Deng - http://creativecommons.org/licenses/by-nc- nd/3.0/		
Download date	2024-05-08 05:32:26		
Item downloaded from	https://hdl.handle.net/10468/1398		



University College Cork, Ireland Coláiste na hOllscoile Corcaigh Ollscoil na hÉireann

Control Circuits for Avalanche Photodiodes

A thesis presented to the National University of Ireland, Cork in Fulfillment for the Requirements for the Degree of Doctor of Philosophy

by

Shijie Deng

Supervisor: Dr. Alan P. Morrison

Head of Department: Prof. Nabeel A. Riza





Department of Electrical and Electronic Engineering, National University of Ireland, Cork

December 2013

Abstract

Avalanche Photodiodes (APDs) have been used in a wide range of low light sensing applications such as DNA sequencing, quantum key distribution, LI-DAR and medical imaging. To operate the APDs, control circuits are required to achieve the desired performance characteristics. This thesis describes the development of three control circuits including a bias circuit, an active quench and reset circuit and a gain control circuit all of which are used for control and performance enhancement of the APDs.

The bias circuit is used to bias planar APDs for operation in both linear and Geiger modes. The circuit was designed and simulated using AMS 0.35 μ m and L-Foundry 0.15 μ m CMOS process. The circuit is based on a dualrail charge pump configuration and operates from a 5 V supply. Simulations shows that it is capable of providing milliamp load currents for shallow-junction planar APDs that operate up to 40 V. Novel voltage regulators allow the bias voltage provided by the circuit to be accurately controlled and easily adjusted by the end user. The circuit is highly integrable requiring only 2 external capacitors thereby enabling a single package APD solution that operates from a single system supply (e.g. 3.3 V, 5 V). This is a very attractive solution for applications requiring a compact integrated APD device.

The active quench and reset circuit is designed for APDs that operate in Geiger-mode and are required for photon counting. The circuit was designed and simulated using L-Foundry 0.15 μ m and AMS 0.35 μ m CMOS process. The chip was fabricated with AMS 0.35 μ m CMOS process. The circuit enables

linear changes in the hold-off time of the Geiger-mode APD (GM-APD) from several nanoseconds to microseconds with a stable step size of 6.5 ns. This facilitates setting the optimal 'afterpulse-free' hold-off time for any GM-APD via user-controlled digital inputs. In addition this circuit doesn't require an additional monostable or pulse generator to reset the detector, thus simplifying the circuit. Compared to existing solutions, this circuit provides more accurate and simpler control of the hold-off time while maintaining a comparable maximum count-rate of 35.2 Mcounts/s.

The third circuit designed is a gain control circuit. This circuit is based on the idea of using two matched APDs to set and stabilize the gain. The circuit was designed and simulated using L-Foundry 0.15 μ m high voltage CMOS process and implemented with discrete components (PCB). Experimental results show that the circuit can provide high bias voltage (up to 40 V) for operating the planar APD, precisely set the APDs gain (with errors of less than 3%) and compensate for changes in the temperature to maintain a more stable gain. The circuit operates without the need for external temperature sensing and control electronics, thus lowering the system cost and complexity. It also provides a simpler and more compact solution compared to previous designs.

This thesis presents the circuit design, simulations and development of the three circuits. These circuits were developed independently of each other and are used for improving different performance characteristics of the APD. Further research on the combination of the three circuits will produce a more compact APD-based solution for a wide range of applications.

Acknowledgments

This thesis would not have been completed without the help of many people. First of all, I would like to thank Dr. Alan Morrison, my project Supervisor, for his excellent guidance and encouragement throughout this research. It was a pleasure to work with him.

Thanks to Dr. John Hayes and Declan Gordon for the useful suggestions on many experimental setups. Thanks also to Jason Hannon and Francesco Pini for their many helpful advice on using Cadence and IC design. I must thank Michael O'Shea in the Mechanical workshop who developed many hardware pieces which made the experimental work easier to complete.

The IT and administrative staff from both Tyndall and the Electrical and Electronic Engineering Department deserve a big thank you for the excellent assistance.

I must thank the friends I made in Cork for so many relaxing weekends.

Finally, I would like to thank my parents who have always supported me and who I know are always there for me.

Publications

• S. Deng and A. P. Morrison, "Integrable bias solution for avalanche photodiodes", Electronics Letters, vol. 47, no. 4, pp. 240-241, Feb. 2011.

• Shijie Deng and Alan P. Morrison, "A dual-rail charge pump bias circuit for avalanche photodiodes", The 22nd Irish Signals and Systems Conference (ISSC 2011), pp. 56-58, Dublin, Ireland, Jun. 2011.

• S. Deng and A. P. Morrison, "A compact bias and gain control solution for avalanche photodiodes", The 24th IEEE Photonics Society Annual Meeting (IPC11), pp. 282-283, Arlington, USA, Oct. 2011.

• Shijie Deng and Alan P. Morrison, "High-resolution hold-off time control circuit for Geiger-mode avalanche photodiodes", The 19th International Conference Mixed Design of Integrated Circuits and Systems (MIXDES2012), pp. 269-272, Warsaw, Poland, May. 2012.

• Shijie Deng and Alan P. Morrison, "Design of an adjustable bias circuit using a single-sided CMOS supply for avalanche photodiodes", The 10th IEEE International NEWCAS Conference (NEWCAS2012), pp. 65-68, Montreal, Canada, Jun. 2012. • Shijie Deng, John M. Hayes and Alan P. Morrison, "A bias and control circuit for gain stabilization in avalanche photodiodes", The 23rd Irish Signals and Systems Conference (ISSC2012), Maynooth, Ireland, Jun. 2012.

• Shijie Deng and Alan P. Morrison, "Active quench and reset integrated circuit with novel hold-off time control logic for Geiger-mode avalanche photodiodes", Optics Letters, vol. 37, no. 18, pp. 3876-3878, Sep. 2012.

• Shijie Deng and Alan P. Morrison, "Design of a Hold-off Time Control Circuit for Geiger-mode Avalanche Photodiodes", International Journal of Microelectronics and Computer Science, vol. 3, no. 2, pp. 53-59, 2012.

Contents

Α	bstra	nct					i
A	ckno	wledgr	nents				iii
\mathbf{P}	ublic	ations					iv
\mathbf{Li}	st of	Figur	es				xii
\mathbf{Li}	st of	Table	S			2	xiii
1	Inti	roduct	ion				1
	1.1	Introd	luction		•		1
	1.2	Thesis	s Outline		•	•	4
2	Inti	roduct	ion to avalanche photodiodes (APDs)				11
	2.1	Photo	detection		•	•	11
	2.2	Avala	nche photodiodes				15
		2.2.1	Avalanche effect in APDs		•		16
		2.2.2	Impact ionisation		•		17
		2.2.3	Multiplication gain		•		20
	2.3	APD	operation modes				22
		2.3.1	Linear mode				22
		2.3.2	Geiger-mode				26
	2.4	Plana	r APD		•		30

	2.5	Conclusion	32
3	Bias	sing Avalanche photodiodes	38
	3.1	Introduction to APD Bias solutions	38
	3.2	Design of an integrable bias circuit	42
		3.2.1 Circuit description	42
		3.2.2 Simulations	48
		3.2.3 Printed circuit board (PCB) implementation	52
	3.3	New approach to the bias circuit $\ldots \ldots \ldots \ldots \ldots \ldots$	54
		3.3.1 Circuit description	55
		3.3.2 Layout and simulations	58
	3.4	Conclusions	63
4	Act	ive quench and reset circuit (AQRC) for Geiger-mode avala	nche
ph	notod	liodes	70
	4.1	Quenching circuits and hold-off control techniques	71
	4.2	High-resolution hold-off time control circuit	79
		4.2.1 Circuit Design	79
		4.2.2 Layout and simulations	82
	4.3	Improved approach and ASIC development of the active quench	
		and reset circuit	86
		4.3.1 Circuit description	86
		4.3.2 ASIC development and measurements	88
	4.4	Conclusion	93
5	Bias	s and gain control circuit for avalanche photodiodes	102
	5.1	Introduction to gain control circuit for	
		avalanche photodiodes	103
	5.2	Circuit description	107
		5.2.1 Charge pump DC-DC converter	108

		5.2.2 Gain control and stabilization	109
	5.3	Simulations	110
	5.4	Development of PCB module and experimental results	112
	5.5	Conclusions	116
~	G		100
6	Con	clusions and future work	123
	6.1	Conclusions	123
	6.2	Future work	124

List of Figures

2.1	Photon absorption mechanisms	12
2.2	Optical absorption in a semiconductor [1]. \ldots \ldots \ldots \ldots	14
2.3	Example of an avalanche photodiode. (a) Diode structure. (b)	
	Doping profile. (c) Energy band diagram. (d) Electric field	
	profile. [1]	17
2.4	(a) Impact ionisation process with electrons only contributing	
	(b) Impact ionisation process with both holes and electrons con-	
	tributing; note the avalanche direction indicated by the blue	
	arrow [1]	18
2.5	Signal and Noise versus gain [19]	25
2.6	Gain variations of APD as a function of bias voltage and tem-	
	perature [20]	26
2.7	Photocurrent when APD operates in (a) linear mode and (b)	
	Geiger-mode.	27
2.8	Cross section of planar shallow junction APD developed by the	
	Photodetection and Imaging Group at University College Cork	
	[2, 33, 34]	31
3.1	Bench top power supply (Keithley 2400 digital source meter) $\ .$.	39
3.2	Commercial bias solutions for APDs: (a) MAXIM MAX5026,	
	5 V to 71 V APD power supply [14] and (b) LINEAR TECH-	
	NOLOGY LT3571, 5 V to 45 V APD bias power supply $[15]$	40

3.3	Basic schematic of a boost converter	41
3.4	Block diagram of overall bias solution	43
3.5	Schematic of dual-rail charge pump	43
3.6	Positive charge pump	44
3.7	Negative charge pump	44
3.8	Schematic of shunt regulator	45
3.9	Schematic of two-stage op-amp	46
3.10	AC sweep of the 2-stage op-amp	47
3.11	Regulation of positive output to 15 V	48
3.12	Regulation of negative output to -15 V \ldots	49
3.13	Simulation results when the bias is regulated to 25 V $\ . \ . \ .$.	50
3.14	Simulation results when the bias is regulated to 30 V $\ . \ . \ .$.	50
3.15	Simulation results when the bias is regulated to 45 V $\ .$	51
2 16	Simulation results of maximum load current and output power	
5.10	Simulation results of maximum load current and output power	
5.10	for different bias voltages	51
3.17	for different bias voltages	51
3.17	for different bias voltages	51 52
3.103.173.18	for different bias voltages	51 52 53
3.103.173.183.19	for different bias voltages	51 52 53 53
3.103.173.183.193.20	for different bias voltages	 51 52 53 53 55
 3.10 3.17 3.18 3.19 3.20 3.21 	for different bias voltages	 51 52 53 53 55 56
 3.10 3.17 3.18 3.19 3.20 3.21 3.22 	for different bias voltages	 51 52 53 53 55 56 56
 3.10 3.17 3.18 3.19 3.20 3.21 3.22 3.23 	for different bias voltages	 51 52 53 53 55 56 56 56 58
3.17 3.18 3.19 3.20 3.21 3.22 3.23 3.24	for different bias voltages	 51 52 53 53 55 56 56 58 59
3.17 3.18 3.19 3.20 3.21 3.22 3.23 3.24 3.25	for different bias voltages	 51 52 53 53 55 56 56 58 59
3.17 3.18 3.19 3.20 3.21 3.22 3.23 3.24 3.25	for different bias voltages	 51 52 53 53 55 56 56 58 59 60
3.17 3.18 3.19 3.20 3.21 3.22 3.23 3.24 3.25 3.26	for different bias voltages	 51 52 53 53 55 56 56 58 59 60

3.27	Positive and negative output when the bias voltage is regulated	
	to 30 V \ldots	61
3.28	Bias voltages ripple and estimated chip size for different load	
	capacitors	62
4.1	Passive quenching circuit	71
4.2	Typical structure of an active quench and reset circuit $\ . \ . \ .$	73
4.3	Diagram of AQRC in [21]	74
4.4	Schematic of a monostable	75
4.5	Modified monostable proposed in $[23]$	75
4.6	Schematic of AQRC reported in [29] and [30]	76
4.7	Diagram of AQRC in [33]	77
4.8	Diagram of delay line technique in [33]	78
4.9	Block diagram of high-resolution hold-off time control circuit	80
4.10	6-bit synchronous binary counter	81
4.11	Layout of the proposed circuit	82
4.12	Simulation model of the GM-APD	83
4.13	Example of the circuit operation when the external inputs are	
	set to 30 ("011110")	84
4.14	Hold-off times versus external input codes when the step reso-	
	lution is set to 2 ns	84
4.15	Setting range of the hold-off time for different step resolutions $% f(x)=\int dx dx dx$.	85
4.16	Block diagram of the new approach	86
4.17	Photograph of the fabricated chip	88
4.18	Top-side of the test PCB (holds the fabricated chip)	89
4.19	Bottom-side of the test PCB (holds the tested APDs)	89
4.20	Experimental setup for testing the AQR-IC	90

4.21	a) Quenching pulse (Qp) and APD's cathode voltage $(V_{cathode})$
	with a hold-off time of 190 ns (00011101), (b) $V_{cathode}$ with hold-
	off time = 326 ns (00110010) and (c) $V_{cathode}$ with hold-off time
	= 1.18 μ s (10110101)
4.22	External input codes versus hold-off time
4.23	Plot of $V_{cathode}$ for saturated count-rate $\ldots \ldots \ldots \ldots \ldots $ 92
5.1	Temperature control system in [15, 16]
5.2	Gain stabilisation system in [8, 17]
5.3	APD bias arrangement using two matched devices 106
5.4	APD gain control and stablilization circuits in [22] 107
5.5	Block diagram of proposed circuit
5.6	Charge pump DC-DC converter
5.7	Schematic of gain control circuit with two matched APDs. $\ . \ . \ . \ 109$
5.8	APD model and its simulation result
5.9	Gain and the bias voltage of APD2 when its gain is set to $100.\ .\ 111$
5.10	Photograph of the fabricated bias and gain control circuit gain. 112
5.11	Output voltages of charge pump DC-DC converter for different
	load currents
5.12	Packaged two matched avalanche photodiodes
5.13	Block diagram of the experimental setup
5.14	Gain variation over a period of time under the control of the
	proposed circuit: (a) Gain set to 20 ; (b) Gain set to 50 ; (c)
	Gain set to 100
5.15	APD's performance when ambient temperature is changed 116
6.1	Single package APD solution with bias circuit

List of Tables

2.1	Characteristics of various semiconductors [1]	13
4.1	Available hold-off times with AQRC in [33]	79
4.2	Comparison of existing active quench and reset circuits \ldots .	95

Chapter 1

Introduction

1.1 Introduction

Advances in avalanche photodiode (APD) technology including higher sensitivity, faster response time, lower operation voltage and lower costs have allowed APDs to replace PIN photodiodes or photomultiplier tubes (PMTs) in many low-light sensing applications.

In fibre communication systems, APDs have been the frequently choices as they can provide higher sensitivity in optical receivers than PIN photodiodes. Thanks to the high internal gain and fast response time, APD can achieve 5~10 dB better sensitivity than PINs and provide higher signal to noise ratio (SNR) that make it more suitable for long haul communications as a high speed receiver in high bandwidth applications [1-3]. In single-photon counting and time-correlated single-photon counting (TCSPC) applications, APDs have been alternatives to photomultiplier tubes (PMTs). APDs can provide higher detection efficiency meanwhile give an excellent photon timing performance. More importantly, the APDs fabricated in planar technology compatible with CMOS circuits have enabled smaller size, lower operating voltage and lower power dissipation than the use of PMTs [4-6]. APDs are also well suited in some other applications including astronomy [7], fluorescence correlation spectroscopy [8], DNA sequencing [9, 10], quantum cryptography [11], LIDAR [12-14], laser range finders [15, 16], metrology [17], confocal microscopy [18], particle detection [19] and nuclear medicine [20, 21].

Avalanche photodiodes operate in different modes under different bias voltages. The APD operates in linear mode with the bias voltage below its breakdown voltage. In this mode, APDs have their performance characterized by gain, noise, and bandwidth. The APD operates in Geiger-mode or single photon mode with the bias voltage above its breakdown voltage and in this mode have their performance characterized by dark count, hold-off time, afterpulsing, response time, jitter and responsivity. Performance characteristics such as quantum efficiency and spectral response are common to both modes of operation. To operate an APD in low light sensing applications, control circuits are required to bias it and also to optimize the APD to achieve the desired performance characteristics.

A number of control circuits have been designed by research groups to operate the APDs and improve their performance. S. Cova *et al* have developed a number of quenching circuits for optimising the performance of GM-APDs. Those works were described in [22-24]. J. Kataoka *et al.* have developed a radiation detector consisting of APDs, an active gain control system and readout circuitry that has been used in a satellite to monitor the distribution of low energy particles [25, 26]. D. Cronin and D. P. O'Connell have also developed some control circuits for the APDs developed by the Photodetection and Imaging Group at University College Cork [27]. These circuits allow the user control of the APD characteristics such as gain and hold-off time [28, 29]. In some commercial organizations, APD control circuits have also been developed. At Hamamatsu, circuits consisting of photodiodes and custom signal processing circuits have been developed which can be used for X-ray detection [30]. PerkinElmer have developed APD/preamplifier hybrid modules which are used for confocal microscopy, range finding and LIDAR applications [31]. Some other organizations such as PicoQuant [32], ID Quantique [33] and Sensl [34] have also developed a number of systems that consist of APDs and control electronics for low light detection.

In this work, the design and development of three APD control circuits is described. These circuits were developed independently of each other. However the combination of the three circuits results in a system that is capable of providing a more compact APD-based solution for a wide range of applications.

The first circuit designed is a bias circuit for an APD that can operate in both linear mode and Geiger-mode. The circuit was designed using AMS 0.35 μ m and L-Foundry 0.15 μm CMOS process and post-layout simulations are presented. Simulation results show that the circuit is capable of sourcing milliamp range load currents for shallow-junction planar APDs that operate up to 40 V. The layout of the circuit designed with the L-Foundry 0.15 μ m process shows a small footprint of $1.55 \text{ mm} \times 1 \text{ mm}$, making it suitable for hybrid integration with two external capacitors and an APD in a single package to operate seamlessly from a 5 V supply. This removes the requirement for a separate APD bias supply, employing the same supply for both signal conditioning and bias. Moreover, the circuit is capable of full integration for APD-based devices. Compared with the existing solutions including external high-voltage supplies, DC-DC converter chips and other reported designs (requiring several external components to function correctly), this design provides a more compact solution that can be used to reduce size, weight and power consumption for APD-based applications.

The second circuit designed is an active quench and reset circuit (AQRC) for the APD that operates in Geiger-mode. This circuit is designed with the aim of overcoming the difficulties of hold-off time adjustment that remain in existing quench circuits. The circuit is simulated and layout designed in the Cadence design environment with L-Foundry 0.15 μ m and AMS 0.35 μ m CMOS process. The integrated circuit was fabricated using AMS 0.35 μ m CMOS process and tested with a GM-APD. Results show that this design is capable of linear changes to the hold-off time from several nanoseconds to microseconds with a reasonable step of 6.5 ns that allows setting of the optimal 'afterpulse-free' hold-off time for any GM-APD through digital inputs or additional signal processing circuitry. A minimum dead time of 28.4 ns was observed from the measurement, demonstrating a saturated photon-counting of 35.2 Mcounts/s with this AQRC.

The third circuit designed is a gain control circuit for APDs that operate in linear-mode. This circuit uses a matched APDs arrangement which has the advantages of controlling and stabilizing the gain without the need for external temperature sensing and control electronics thereby lowering the system cost and complexity. Meanwhile this design provides a simpler and more compact solution. The circuit was simulated in the Cadence environment with L-Foundry 0.15 μ m high voltage CMOS process, fabricated with discrete components (PCB) and measured with packaged APDs. Results show that the circuit can provide in excess of 40 V bias voltage for the planar APD and allow for the setting and stabilization of its gain independent of variation in the ambient temperature or power supply fluctuation.

1.2 Thesis Outline

This thesis is divided into a number of chapters. Chapter 2 introduces the theory of photodetection and the operation of avalanche photodiodes biased in linear mode and Geiger-mode. Chapter 3 describes a bias solution that is used for APDs operating in both linear and Geiger modes. Chapter 4 presents the design and development of a new active quench and reset circuit (AQRC) for Geiger-mode avalanche photodiodes (GM-APDs). Chapter 5 describes a novel gain control and stabilization circuit for avalanche photodiodes (APDs) biased in linear mode. In Chapter 6, the conclusions of this thesis are made and future work is proposed.

Bibliography

- Joe C. Campbell, "Recent Advances in Telecommunications Avalanche Photodiodes", IEEE Journal of Lightwave Technology, Vol.25, No.1, pp.109-121, 2007.
- [2] Solomon Assefa, Fengnian Xia and Yurii A. Vlasov, "Reinventing germanium avalanche photodetector for nanophotonic on-chip optical interconnects", Nature, Vol.464, No.7258, pp.80-84, Mar. 2010.
- [3] Osayd Kharraz and David Forsyth, "Performance comparisons between PIN and APD photodetectors for use in optical communication systems", International Journal for Light and Electron Optics, Vol.124, No.13, pp.1493-1498, Jul. 2013.
- [4] A. M. Wallace, G. S. Buller, and A. C. Walker, "3D imaging and ranging by time-correlated single photon counting", IEEE Journal of Computing and Control Engineering, Vol.12, No.4, pp.157-168, Aug. 2001.
- [5] Massimo Ghioni, Angelo Gulinatti, Ivan Rech, Franco Zappa and Sergio Cova, "Progress in Silicon Single-Photon Avalanche Diodes", IEEE Journal of Selected Topics in Quantum Electronics, Vol.13, No.4, pp.852-862, Jul./Aug. 2007.
- [6] F. Villa, B. Markovic, S. Bellisai, D. Bronzi, A. Tosi, F. Zappa, S. Tisa,D. Durini, S. Weyers, U. Paschen and W. Brockherde, "SPAD Smart

Pixel for Time-of-Flight and Time-Correlated Single-Photon Counting Measurements", IEEE Photonics Journal, Vol.4, No.3, pp.795-804, Jun. 2012.

- [7] M. Martinez, A. Ostankov, E. Lorenz, R. Mirzoyan, and T. Schweizer, "Contraints in using APDs in air Cherenkov telescopes for y-astronomy", Nuclear Instruments and Methods in Physics Research A, Vol.442, No.1-3, pp.209-215, Mar. 2000.
- [8] M. Oda, Y. Yamashita, E. Ohmae, and Y. Tsuchiya, "Tissue oxygenation monitoring with a time-resolved spectroscopy system", CLEO Pacific Rim Conference on Lasers and Electro-Optics, pp.378-379, Jul. 2001.
- [9] S. Fore, T. A. Laurence, Y. Yeh, R. Balhorn, C. W. Hollars, M. Cosman, and T. Huser, "Distribution analysis of the photon correlation spectroscopy of discrete numbers of dye molecules conjugated of DNA", IEEE Journal of Selected Topics in Quantum Electronics, Vol.11, No.4, pp.873-880, Jul. 2005.
- [10] S. A. Soper, J. H. Flanagan, B. L. Legendre, D. C. Williams, and R. P. Hammer, "Near-infrared, laser-induced fluorescence detection for DNA sequencing applications", IEEE Journal of Selected Topics in Quantum Electronics, Vol.2, No.4 pp.1129-1139, Dec. 1996.
- [11] G. Ribordy, N. Gisin, O. Guinnard, and H. Zbinden, "Single-photon detection at telecom wavelength and quantum cryptography", Annual Meeting of the IEEE Lasers and Electro-Optics Society, pp.771-772, 2003.
- [12] B. Dion, P. Iepage, C. Iedoux, and N. Bertone, "High performing photodiodes for demanding applications", LEOS Summer Topical Mettings, pp.48-49, Jul. 2006.

- [13] T. Maruyama, F. Narusawa, M. Kudo, M. Tanaka, Y. Saito, T. Yoshida, and A. Nomura, "A near-infrared photon-counting system using an In-GaAs avalanche photodiode", Conference on Lasers and Electro-Optics, pp.138-139, May 2000.
- [14] Y. Kawamura and T. Imai, "Development of spaceborne LIDAR: ELISE (Experi-mental Lidar In Space Equipment)", CLEO/Pacific Rim Conference on Lasers and Electro-Optics, pp.270271, Aug. 1999.
- [15] J. S. Massa, M. Umasuthan, A. M. Wallace, G. S. Buller and A. C. Walker, "Rangefinding using time correlated single photon counting", IEEE International Conference on Recent Advances in 3-D Imaging and Modeling, pp.36-43, May 1997.
- [16] Emmanuel R. Moutaye and Hlne Tap-Bteille, "CMOS Avalanche Photodiode Embedded in a Phase-Shift Laser Rangefinder", IEEE Transactions on Electron Devices, Vol.55, No.12, Dec. 2008.
- [17] T. Miyata, T. Araki, and T. Iwata, "Correction of the intensity-dependent phase delay in a silicon Avalanche Photodiode by controlling its reverse bias voltage", IEEE Journal of Quantum Electronics, Vol.39, No.7, pp.919-923, Jul. 2003.
- [18] Y. Zhang and A. Roorda, "Photon signal detection and evaluation in the adaptive optics scanning laser ophthalmoscope", Journal of the Optical Society of America, Vol.24, No.5, pp.1276-1283, May 2007.
- [19] S. M. Khanna, S. Charbonneau, P. G. Piva, M. Parenteau, and C. Carlone, "Effects of 3 MeV proton irradiation on the excitonic lifetime in gallium arsenide", IEEE Transactions on Nuclear Science, Vol.45, No.1, pp.2430-2435, Dec. 1998.

- [20] J.-F. Pratte, S. Robert, G. De Geronimo, P. O'Connor, S. Stoll, C. M. Pepin, R. Fontaine, and R. Lecomte, "Design and performance of 0.18um CMOS charge preamplifiers for APD-based PET scanners", IEEE Transactions on Nuclear Science, Vol.51, No.5, pp.1979-1985, Oct. 2004.
- [21] M. C. Maas, D. J. van der Laan, D. R. Schaart, J. Huizenga, J. C. Brouwer, P. Bruyndonchx, S. Leonard, C. Lemaitre, and C. W. E. van Eijk, "Experimental characterization of monolithic-crystal small animal PET detectors read out by APD arrays", IEEE Transactions on Nuclear Science, Vol.53, No.3, pp.1071-1077, Jun. 2006.
- [22] S. Cova, M. Ghioni, A. Lacaita, C. Samori and F. Zappa, "Avalanche photodiodes and quenching circuits for single-photon detection", Applied optics, Vol.35, pp.1956-1976, 1996.
- [23] F. Zappa, M. Ghioni, S. Cova, C. Samori and A. Giudice, "An integrated active-quenching circuit for single-photon avalanche diodes", IEEE Transactions on Instrumentation and Measurement, Vol.49, pp.1167-1175, 2000.
- [24] F. Zappa, A. Lotito, A. Giudice, S. Cova and M. Ghioni, "Monolithic active-quenching and active-reset circuit for single-photon avalanche detectors", IEEE Journal of Solid-State Circuits, Vol.38, pp.1298-1301, 2003.
- [25] J. Kataokaa, R. Satoa, T. Ikagawaa, J. Kotokua, Y. Kuramotoa, Y. Tsubukua, T. Saitoa, Y. Yatsua, N. Kawaia, Y. Ishikawab and N. Kawabatab, "An active gain-control system for Avalanche photodiodes under moderate temperature variations", Nuclear Instruments and Methods in Physics Research A, Vol.564, No.1, pp.300-307, 2006.
- [26] J. Kataoka, T. Toizumi, T. Nakamori, Y. Yatsu, Y. Tsubuku, Y. Kuramoto, T. Enomoto, R. Usui, N. Kawai, H. Ashida, K. Omagari, K. Fujihashi, S. Inagawa, Y. Miura, Y. Konda, N. Miyashita, S. Matsunaga,

Y. Ishikawa, Y. Matsunaga and N. Kawabata, "In-orbit performance of avalanche photodiode as radiation detector on board the picosatellite Cute-1.7+APD", Journal of Geophysical Research, Vol.115, A05204, May 2010.

- [27] http://www.ucc.ie/en/eleceng/research/researchareas/photodectect/
- [28] Diarmuid Patrick O'Connell, "Control circuit for Avalanche Photodiodes", Ph.D. thesis, University College Cork, Cork, Ireland, Feb. 2009.
- [29] Donal Cronin, "Automatic Control Circuits for Geiger-Mode Avalanche Photodiodes", Ph.D. thesis, University College Cork, Ireland, Mar. 2009.
- [30] https://www.hamamatsu.com/jp/en/product/category/3100/4005/ 4122/index.html.
- [31] http://www.perkinelmer.co.uk/PDFS/downloads/ cat_sensorsandemittersphotondetection.pdf
- [32] http://www.picoquant.com/
- [33] http://www.idquantique.com/
- [34] http://sensl.com/

Chapter 2

Introduction to avalanche photodiodes (APDs)

This chapter briefly introduces the theory of photodetection and the operation of avalanche photodiodes (APDs). The APD's physical processes and key performance characteristics are explained. The backgrounds of some external control circuits required for the APDs' operation are described. The design of the planar APD used throughout this project is illustrated.

2.1 Photodetection

Based on photon absorption in semiconductor materials, a photodetector absorbs light via three processes including intrinsic band-to-band absorption, free carrier absorption and band-to-impurity absorption, see Figure 2.1. Intrinsic band-to-band absorption occurs when the photon energy $(h\nu)$ is greater than the material bandgap energy E_g . In this case, a photon is able to excite an electron from the valence band up into the conduction band to create an electron-hole pair. The hole and electron each constitute a charge carrier. The electric field applied to the semiconductor will cause the hole and electron to be transported into the external circuit through the material. When the electron-hole pair makes its way to the external circuit their combined effect will lead to a flow of charge q in the circuit where q is the charge of a single electron [1], [2].



Figure 2.1: Photon absorption mechanisms.

Intrinsic band-to-band absorption is the dominant absorption mechanism in most semiconductors used for photodetection. To create an electron-hole pair, it requires that the photon energy, $\mathbf{E} = h\nu$, is greater than the material bandgap energy, E_q : this can be expressed as

$$hv > E_g \text{ or } h \frac{c}{\lambda} > E_g$$

$$(2.1)$$

This yields a maximum allowable wavelength

$$\lambda_{max} = \frac{hc}{E_g} \text{ or } \lambda_{max} (nm) = \frac{1240}{E_g (eV)}$$
(2.2)

Where E_g (eV) = bandgap energy in electron-volts.

Table 2.1 lists the bandgap, the corresponding maximum usable wavelength, and typical operating wavelengths for common semiconductor materials.

Free carrier absorption occurs when the photon energy is absorbed by free carriers in either the conduction band or the valence band which corresponds to the "heating" of the semiconductor material. It is a secondary effect at the near infrared wavelengths used for optical communications. Band-to-impurity

Material	Bandgap	Maximum wave-	Typical operat-
		length (nm)	ing range (nm)
Si	1.12	1110	500 - 900
Ge	0.67	1850	900 - 1300
GaAs	1.43	870	750 - 850
$In_xGa_{1-x}As_yP_{1-y}$	0.38 - 2.25	550 - 3260	1000 - 1600

Table 2.1: Characteristics of various semiconductors [1].

absorption is another secondary effect at near-IR wavelengths. It is used to construct photodetectors responsive at mid-IR wavelengths as long as 30 μ m [1].

An important parameter for a photodiode is the quantum efficiency (QE). The quantum efficiency is defined as the probability that a single photon incident on the detector generates an electron-hole pair and is given by [1]:

$$\eta = \frac{number \, of \, e \, -h \, pairs \, produced}{number \, of \, incident \, photons} \tag{2.3}$$

where $0 \le \eta \le 1$. A signal with photon energy sufficient to generate photocarriers will continuously lose energy to the semiconductor crystal lattice as the optical field propagates through the semiconductor. As can be seen from Figure 2.2, there is a reflection loss at the interface of the air and the semiconductor due to differences in the index of refraction. The Fresnel reflectivity for an optical signal at normal incidence to an interface between two materials is given by

$$R = \frac{(n_1 - n_2)^2}{(n_1 + n_2)^2} \tag{2.4}$$

where $n_1 =$ index of refraction of first material and $n_2 =$ index of refraction of second material. For many semiconductors used in the 800 nm to 1500 nm region, n is between 3.2 and 3.6 and the reflectivity can be as high as 32 %. Reflection losses can be reduced with antireflection coatings to less than 1 %.

Inside the semiconductor, the field decays away exponentially as energy is transferred to the semiconductor. The material can be characterised by an



Figure 2.2: Optical absorption in a semiconductor [1].

absorption constant α and a penetration depth $1/\alpha$. Penetration depth is the point at which 1/e of the optical signal power remains. The power in the optical field decays with distance. The amount of power absorbed in the semiconductor is as a function of position within the material and can be expressed as

$$P_{abs} = P_i (1 - R) \left(1 - e^{-\alpha x} \right)$$
(2.5)

The number of photons absorbed per second is the power in watts divided by the photon energy ($\mathbf{E} = h\nu$). If each absorbed photon generates a pair of photocarriers (i.e. an electron-hole pair), the number of photocarriers generated per photon for a specific semiconductor with reflectivity R and absorption constant α is given by

$$\eta(x) = (1 - R) \left(1 - e^{-\alpha x} \right)$$
(2.6)

Where $\eta(x)$ is the quantum efficiency of the photodetector.

Responsivity is defined as the ratio of the photocurrent to the optical power and is given by [2]

$$R = \frac{I_{photo}}{P_{opt}} = \frac{\eta q}{hv} \left(A/W \right) \tag{2.7}$$

Where I_{photo} = photogenerated current, P_{opt} = incident optical power, η = quantum efficiency, q = electronic charge, h = Planck's constant, v = optical frequency. For an ideal photodiode each incident photon would generate the charge of one electron flowing in the external circuit. In reality, there are several physical effects that tend to reduce the responsivity, including incomplete absorption, recombination, reflection from the semiconductor surface and contact shadowing.

The responsivity of a semiconductor will also vary with wavelength [1]. Responsivity increases as the wavelength increases because there are more photons per watt at long wavelengths than there are at short wavelengths. This is a direct result of the photon energy decreasing with wavelength. Since the amount of photocurrent is determined by the number of photons instead of the energy of the photons, longer wavelengths generate more photocurrent per watt than the short wavelengths.

2.2 Avalanche photodiodes

An avalanche photodiode (APD) is a very sensitive photodetector. The advances in APDs such as low operating voltages, low-cost, higher sensitivity and small-size make it very attractive for use in photodetection [1]. It has been used in a wide range of lowlight sensing applications including astronomy [3], DNA sequencing [4], light detection and ranging (LIDAR) [5] and medical sensing [6].

2.2.1 Avalanche effect in APDs

In Figure 2.3, the structure of a typical avalanche photodiode is illustrated [1]. The APD consists of four sections of semiconductor with different doping profiles as shown in Figure 2.3(b). The leftmost section is a p-type semiconductor doped to a density of N_{accept} , while the rightmost section is an n-type semiconductor doped to a density N_{donor} . The middle region is an intrinsic region in combination with a second p-type region. The energy diagram of the APD is shown in Figure 2.3(c) while the electric field profile of the space charge region is shown in Figure 2.3(d). When a p-n junction is subjected to a high reverse bias there are two breakdown mechanisms that can occur. The first mechanism involves the atoms being directly ionized by the applied field which is known as zener breakdown and is commonly used to make voltage regulating diodes. The second mechanism is called avalanche breakdown which is caused by high velocity carriers causing impact ionisations within the semiconductor that generates additional carriers.

In the absorption region shown in Figure 2.3(d), photons are absorbed in the lightly doped semiconductor material, generating electron-hole pairs. The carriers rapidly travel through the lower field drift region at their saturation velocity $(1 \times 10^7 \text{ cm/s} \text{ for electric field} > 10^4 \text{ V/cm [7]})$ to the edges of the region. Carriers that reach the gain region are further accelerated by an even higher electric field. Avalanche breakdown then begins to occur with very high velocity carriers causing impact ionisations within the semiconductor that will further generate additional carriers. These additional carriers can then undergo additional impact ionisations causing an avalanche effect.



Figure 2.3: Example of an avalanche photodiode. (a) Diode structure. (b) Doping profile. (c) Energy band diagram. (d) Electric field profile. [1]

2.2.2 Impact ionisation

The impact ionisation or multiplication process in the gain region is demonstrated in Figure 2.4. The avalanche current flowing in the APD is caused by this mechanism. The impact ionisation can be characterised by the ionisation coefficients, α_e and α_h , for electrons and holes respectively. α_e and α_h can also be defined as the number of secondary electron-hole pairs generated in a unit distance with the applied electric field. For a given temperature, the ionization coefficients are exponentially dependent on the electric field and can be defined as

$$\alpha_{e,h} = a.exp\left(-\left[\frac{b}{E}\right]^c\right) (cm^{-1})$$
(2.8)

where a, b and c are experimentally determined constants, E = magnitude of the electric field.



Figure 2.4: (a) Impact ionisation process with electrons only contributing (b) Impact ionisation process with both holes and electrons contributing; note the avalanche direction indicated by the blue arrow [1].

If the temperature of the detector is increased, there will be an increase in both

the number of non-ionising collisions with thermally excited atoms and the phonon energy as shown in Eqn. 2.9. The additional phonon collisions reduce the carrier velocity and decreases their probability of gaining sufficient energy to initiate impact ionisations and the generation of the number of secondary electron-hole pairs will be decreased [8-13].

$$\alpha_{e,h} = \frac{q\varepsilon}{E_i} exp\left(-\frac{3E_{phonon}E_i}{\left(q\varepsilon\lambda\right)^2}\right)$$
(2.9)

where q = electronic charge, E = electric field, $E_{phonon} =$ phonon energy, $E_i =$ ionisation threshold energy (minimum energy for impact ionisation to occur), $\lambda =$ mean free path for phonon scattering.

To determine the ionising carrier in an APD, an ionisation ratio is defined as:

$$k = \frac{\alpha_h}{\alpha_e} \tag{2.10}$$

The APD structure is optimised for the more ionising carrier depending on this ratio k. For materials where $\alpha_e \gg \alpha_h$, (i.e. where electrons are much more ionising than holes such as in silicon) it is required that photons are absorbed in the p-region (see Figure 2.3). Values of k between 0.01 and 0.003 can be achieved to produce high quality Si APDs. Similarly, for materials where holes are more ionising than electrons ($\alpha_h \gg \alpha_e$, e.g. Germanium), photons should be absorbed at the n-region of the junction. Figure 2.4(a) shows the impact ionisation process for silicon (k \ll 1), where holes do not contribute. When a photoelectron enters into the gain region, it is accelerated by the applied electric field and then gains enough kinetic energy from the field to generate a series of impact ionisations. In this example, it results in six extra electrons and the avalanche proceeds in a well defined manner from left to right. It is assumed that only holes contribute to the impact ionisations when k \gg 1. Figure 2.4(b) demonstrates an undesirable case where k \cong 1 and both electrons and holes contribute to the impact ionisations. In this example, an electron initiates the first ionisation but the subsequent ionisations can be caused by high energy electrons or holes. This avalanche process behaves in an uncontrolled manner, moving back and forth in the gain region depending on which carrier is causing the ionisation. This leads to an undesirable uncertainty associated with this type of process as there is no clearly defined end point until all possible impact ionisations are completed [1, 2]. The net result of this is an undesirable noise contribution due to the stochastic nature of impact ionisation called excess noise.

2.2.3 Multiplication gain

The gain or multiplication factor of an APD is given by [2, 7]

$$M = \frac{I_{photo}}{I_{primary}} \tag{2.11}$$

Where I_{photo} = observable photocurrent at APD terminals, $I_{primary}$ = internal photocurrent before multiplication. The primary current is defined by

$$I_{primary} = \frac{\eta q}{hv} P_{rcvd} \tag{2.12}$$

Where $\eta =$ quantum efficiency, $P_{rcvd} =$ received optical signal power. Typically, multiplication gains of a few tens to a few hundreds are applied, even though it is possible for the multiplication factor to be as high as 10^3 or 10^4 [1, 14]. The expressions for electron multiplication (M_e) and hole multiplication (M_h) in a gain region of length l_g are [1, 2, 7]

$$M_e = \frac{1}{1 - \int_{l_0}^{l_g} \alpha_e exp\left(-\int_0^x \left(\alpha_e - \alpha_h\right) dx\right) dx}$$
(2.13)

$$M_{h} = \frac{exp\left(-\int_{0}^{l_{g}} \left(\alpha_{e} - \alpha_{h}\right) dx\right)}{1 - \int_{l_{0}}^{l_{g}} \alpha_{e} exp\left(-\int_{0}^{x} \left(\alpha_{e} - \alpha_{h}\right) dx\right) dx}$$
(2.14)

As both α_e and α_h in most devices are functions of position within the gain region, exact solutions to Equations 2.13 and 2.14 are difficult to obtain. In certain cases closed form solutions are available. One is for the case where α_e $= \alpha_h$, only electrons are being injected into the gain region, the electric field in the gain region is uniform and the number of ionising collisions per primary carrier is large. In this case, the electron multiplication is given by

$$M_e = \frac{1 - k}{\exp(-(1 - k)\,\alpha_e l_g) - k}$$
(2.15)

Where $l_g = \text{length of gain region}$. For the case where k = 0, Equation 2.15 can be reduced to

$$M_e = \frac{1}{1 - \alpha_e l_g} \tag{2.16}$$

Instability occurs for $l_g = 1/\alpha_e$ and this condition must be avoided. In a practical device, the maximum achievable DC multiplication is limited by the series resistance and space charge effect (these factors can be combined into a single series resistance R). The multiplication for photogenerated carriers can be described as [2, 15]

$$M_{e} = \frac{I - I_{MD}}{I_{P} - I_{D}} = \frac{1}{\left[1 - \left(\frac{V_{R} - IR}{V_{B}}\right)^{n}\right]}$$
(2.17)

Where I = total multiplied current, I_P = total primary (unmultiplied) current, I_D = primary dark current, I_{MD} = multiplied dark current, V_R = reverse bias voltage, V_B = zener breakdown voltage, n = constant depending on the semiconductor material, doping profile and radiation wavelength. For high light intensity, $I_P \gg I_D$ and $IR \ll V_B$, the maximum value of the multiplication is given by

$$\|\| (M_{ph})_{max} \cong \frac{I}{I_P} = \frac{1}{\left[1 - \left(\frac{V_R - IR}{V_B}\right)^n\right]} | V_R \to V_B \approx \frac{1}{nIR/V_B}$$
(2.18)
$$(M_{ph})_{max} = \sqrt{V_B/nI_PR} \tag{2.19}$$

When the photocurrent is smaller than the dark current, the maximum multiplication is limited by the dark current and is given by an expression similar to Equation 2.19, with I_P replaced by I_D . It is important that the dark current is made as low as possible so that it does not limit $(M_{ph})_{max}$.

2.3 APD operation modes

With different bias voltages, avalanche photodiodes operate in different modes. APDs that are biased below the breakdown voltage (linear mode) have their performance characterized by gain, excess noise, responsivity and bandwidth while APDs biased above the breakdown voltage (Geiger-mode) have their performance characterized by photon detection probability, dark count rate, hold-off time, afterpulsing effect, and response time jitter. Some performance characteristics such as quantum efficiency and spectral response are common to both modes of operation.

2.3.1 Linear mode

Linear mode applies to APDs biased below the breakdown voltage. In linear mode, the current output varies linearly with the incident optical power and the gain to the incident photo-generated carriers can be from one up to several hundred.

The bandwidth associated with avalanche photodiodes is limited by three main factors including RC time constant, transit time and avalanche multiplication build-up time. The RC time constant is determined by the parallel plate capacitor which shunts current away from the external circuit and the junction

or

resistance. The junction capacitance is determined by the APD geometry, the fabrication process and applied bias. Since the duration of the photocurrent depends on the lifetime of the longest lived carrier, any fluctuations in the received optical field faster than the carrier lifetime will not be clearly observed. An overall estimate for the APD bandwidth is given by the root-sum-square of the two individual bandwidths [16].

$$B = \frac{1}{\sqrt{\left(\frac{1}{f_{RC}}\right)^2 + \left(\frac{1}{f_t}\right)^2}} \tag{2.20}$$

Where $f_{RC} = \text{RC}$ bandwidth, $f_t = \text{transit time bandwidth}$. The time delay due to avalanche multiplication build up can also affect APD bandwidth. The bandwidth of an avalanche photodiode has been shown to be independent of avalanche multiplication when the DC multiplication is less than α_e/α_h [17]. In this circumstance the bandwidth is given by Equation 2.20. Other factors that affect response speed are carrier diffusion current from outside the gain region and the space charge effect. The time delay of carriers generated outside the gain region occurs when the absorption region is not wide enough with respect to the penetration depth of the incident light. The space charge effect is evident when the incident light level is high and the resulting photocurrent is large, with the attractive power of the electron-hole pairs diminishing the electric field effect.

Similar to a passive component, the APD generates signal noise that can be characterised into a number of types. Thermal noise is a temperature dependent white noise source. The thermal noise generated by an APD is given by

$$v_R^2 = 4kTR_LB \tag{2.21}$$

Where k = Boltzmann's constant, $T = the absolute temperature, <math>R_L = the$

load resistance, B = the bandwidth. The load resistance is the series resistance connected to the APD as well as the APD resistance itself.

Shot noise is a current dependent white noise source. The total shot noise generated by an APD is given by Equation 2.22:

$$\bar{i^2} = 2qIB \tag{2.22}$$

Where q = electron charge, I = total APD current, B = the bandwidth. The total current is made up of the dark current and the photocurrent. The dark current has a surface leakage component, I_{DS} , that does not flow through the gain region and so does not take part in the multiplication process. On the other hand the internal dark current component, I_{DG} , flow through the gain region and adds to the gain current.

The ionization of individual carriers is not uniform and so the multiplication process contains statistical fluctuations known as excess noise. The excess noise factor, F, can be expressed by the multiplication ratio, M and the ratio of the electron/hole ionization rate, k, and is defined as [16, 18]:

$$F = Mk + \left(2 - \frac{1}{M}\right)(1 - k) \tag{2.23}$$

Equation 2.23 is the noise factor when electrons are injected into the avalanche region. To evaluate the excess noise factor when holes are injected into the avalanche region, k in Equation 2.23 should be substituted by 1/k. In the ideal case noise is minimized, k should equal zero for electron injection and k should be infinite for hole injection. Silicon APDs are usually used when electrons are injected into the avalanche region. When the thermal effects and total current noise are taken into account the final expression for APD noise is given as:

$$\bar{I}_D^2 = 2q \left(I_P + I_{DG} \right) M^2 F B + 2q I_{DS} B + \frac{4kTB}{R_L}$$
(2.24)

Where I_P is the photocurrent for M=1. From Equation 2.24 it can be seen that excess noise increases as the gain increases. The photocurrent generated by the optical signal is also amplified by the gain and as a result there is an optimum gain value which maximizes APD signal to noise performance. This optimum gain is achieved when the shot noise equals the thermal noise, as shown in Figure 2.5 [16, 19].



 F_{AMP} : Noise figure of next stage amplifier R_{in} : Input resistance of next stage amplifier

Figure 2.5: Signal and Noise versus gain [19].

The signal to noise ratio for an APD can be calculated as follows:

$$S/N = \frac{I_P^2 M^2}{2q \left(I_P + I_{DG}\right) B M^2 F + 2q I_{DS} + \frac{4kTB}{R_L}}$$
(2.25)

Operating conditions such as temperature and bias voltage play a large part in determining the APDs performance. The APD gain has a temperature dependent characteristic [16, 20]. The gain at a certain bias voltage decreases with increasing temperature. These effects can be seen in Figure 2.6.

To achieve maximum signal to noise ratio in the APD, it is necessary to set



Figure 2.6: Gain variations of APD as a function of bias voltage and temperature [20].

the APD gain close to the optimum gain and stabilize it. This is accomplished by maintaining a constant temperature or adjusting the bias voltage according to the changes in temperature using appropriate control circuits. Those circuits will be introduced in more detail in Chapter 5 and a new bias and gain control circuit will be presented which can be used to maintain the APD gain independent of environmental conditions.

2.3.2 Geiger-mode

Geiger-mode avalanche photodiodes are capable of operating as single photon counting diodes or single photon avalanche diodes (SPAD). In Geiger-mode, the APD is biased above its breakdown voltage. When a photon is absorbed by the APD, an avalanche event is triggered and registered as a photon count. External circuitry, known as a quench circuit, senses this event and reduces the bias across the APD to allow it to recover. After the current dissipates, the voltage is reset, and the diode can count another avalanche event [21]. In contrast to the APD operating in linear mode, the output of a Geiger-mode



Figure 2.7: Photocurrent when APD operates in (a) linear mode and (b) Geiger-mode.

APD is not a linear current, but current pulses. The difference between operating an APD in the linear-mode and the Geiger-mode can been seen in Figure 2.7, where the photocurrent of the former is proportional to the incident optical power while the light intensity on a Geiger-mode APD gives a corresponding change in the number of current pulses per second [12]. Operating the APD in Geiger-mode eliminates many of the current fluctuations present in the linear mode. The output current pulses are sufficiently high so that variations between pulses do not matter. The external circuitry only detects that a pulse has occurred [21].

Dark count

When counting photons, the GM-APD detects the avalanche events created in the APD. The avalanche breakdowns are not only generated by photon absorption but also by other means such as thermal generation of electron-hole pairs and carriers tunnelling across the depletion region [7, 12, 21, 22, 23]. These pulses result in a noise source for the Geiger-mode detector and are referred to as dark counts due to the detection of events in the absence of a light source. The dark count rate of a GM-APD is an important parameter that should be minimised. Shockley-Reed-Hall (SRH) generation and recombination theory describes the thermal excitation of carriers in semiconductors. If an electronhole pair is thermally generated in the depletion region, an avalanche event may occur as outlined in Section 2.2.1. If these are generated in the p-epilayer region of the GM-APD, these carriers may drift to the depletion region and cause an avalanche [12, 24]. Band-to-band tunnelling contributes to the number of dark counts if the electric-field strength is high $(5 \times 10^5 \text{ V } cm^{-1})$ [24, 25]. However, high electric-fields are required in GM-APDs to maximise the photon detection probability [21]. Due to these high electric-fields, an electron may tunnel from the valence band to the conduction band, thus leaving a hole behind in the valence band. As a result, the electron-hole pair may generate an avalanche event and hence a dark count. When the acceptor concentration of the p-region of the GM-APD's pn junction is increased, i.e. N_a $\geq 5 \times 10^{15} \ cm^{-3}$, and hence the breakdown voltage, V_{br} , is less than 25 V, the field strength in the junction is increased such that the dark count rate due to tunnelling is increased. When operating the GM-APD, the dark count rate will increase as the following parameters increase: excess bias voltage (the bias voltage above the breakdown voltage), detector area and temperature [21, 26-30. Decreasing the excess bias will decrease the dark count but unfortunately the photon detection probability will also be reduced. Decreasing the active area is possible but may not suit all applications such as imaging. The temperature dependence of the dark count can be minimised by cooling the GM-APD using a thermoelectric cooler (TEC) [12].

Afterpulsing

During an avalanche event, some of the charge carriers will be captured by trapping centres in the space charge region when the avalanche current flows through the device and there are released after a period of time. These carriers could trigger a new avalanche event, which in turn can lead to further carrier trapping. The avalanche events triggered by trapped carriers are not related to new photon arrivals and are thus an unwanted source of noise called afterpulsing. Afterpulsing intensity depends on the type and number of trapping centres present in the device and the amount of charge that flows through the junction during an avalanche. Afterpulsing introduces correlation between consecutive avalanches, and limits the maximum repetition rate and so needs to be minimised.

The afterpulsing phenomenon can be minimized by allowing sufficient time for all trapped charge to dissipate before resetting the GM-APD. This is achieved using an appropriate control circuit that quenches the APD avalanche current by lowering the bias voltage below the breakdown voltage, holding the bias below breakdown for a period of time before resetting the device to its original bias voltage to await the next avalanche event. The period of time that the device is held below its breakdown voltage is known as the hold-off time. During the hold-off and reset process, no incoming photon can be detected and this leads to a reduction of the maximum achievable count-rate of the GM-APD. As a result, a trade-off needs to be made between the count-rate and an acceptable level of afterpulsing. The control circuits used to minimize the afterpulsing phenomenon and setting the hold-off time will be discussed in detail in Chapter 3 and a quenching circuit with improved hold-off time control logic for the GM-APD will be introduced.

Jitter

Timing jitter is the delay between a photon being absorbed by the detector and the detector sending out a measurable response. It can be described as the fullwidth half-maximum measure of the temporal variation in the avalanche breakdown pulses resulting from an incident photon. Among the timing-resolution components are the variation caused by the generated carrier transit time from the depletion layer to the multiplication region, which is dependent on the depth of absorption of the incident photon and the statistical build up of the avalanche current itself [21, 26, 32].

2.4 Planar APD

The APD used throughout this research is the planar junction APD developed by the Photodetection and Imaging Group at University College Cork [33, 34]. A cross section of the APD can be seen in Figure 2.8. The APD can be fabricated with CMOS compatible process steps with the p-epitaxy common to the CMOS process that allows the APD to be monolithically integrated with CMOS circuitry. The central p-implant is the enrichment implant that increases the p-doping compared to the epi-layer. The overlap of the p and n layers defines the active area of the detector. The high field region is the depletion region that occurs around the pn junction where the photoelectron impact ionisation takes place. The n+ layer overlaps the high-field region into the p-epitaxial substrate, forming a virtual guard ring. The guard ring eliminates edge breakdown of the diode caused by the junction curvature effects, allowing a uniform breakdown across the surface of the junction. The n+ layer forms the diode's cathode. The annular p+ sinkers allows top contact to the anode which aid in the removal of metallic contaminants from the high-field region [16, 35, 36]. The shallow junction of the planar APD gives the detector increased sensitivity to short wavelength light (400 to 650 nm). The doping of the junction is engineered to allow for low voltage operation (< 50 volts).



Figure 2.8: Cross section of planar shallow junction APD developed by the Photodetection and Imaging Group at University College Cork [2, 33, 34].

Planar APDs have a number of advantages over Large Area APD (LAAPD) bevelled edge structures and reach-through APD structures. In LAAPDs the impact ionisation takes place in the p-region where the electric field is low. It takes substantial time, 10ns-30ns, for multiplication and charge collection to take place. LAAPDs need a large operating voltage of between 1000 and 2000 volts. Cooling systems, such as liquid nitrogen, are needed for LAAPDs to operate in photon counting applications. Reach through APDs have high operating voltages of between 300 - 450 volts. Also, placing the high field region deep inside the detector structure and the use of a pi region requires specialised processing steps. These steps increase fabrication costs and are not CMOS compatible [2, 16]. The planar APD has a simple structure requiring fewer process steps. This leads to fewer defects and higher performance detectors. Moreover, the low operation voltage (< 50 V) makes it possible to design an on-chip bias circuit to power the APD. As will be seen in Chapter 3, an intergrable bias solution is designed for biasing the planar APD that can be used to reduce the size, weight and power consumption of the APD-based applications.

2.5 Conclusion

This chapter has introduced the theory of photodetection and some background detail of APDs that operate in both linear and Geiger modes. The performance characteristics of the APD such as quantum efficiency, multiplication gain, bandwidth, dark count and afterpulsing have been discussed. The APD requires many external control circuits for enhancing its performance. In the following chapters, some of these control circuits will be introduced and improved circuit designs will be presented. These circuits include a bias circuit, a gain stabilization circuit and a quenching circuit. The bias circuit is used to provide the high bias voltage for powering the APDs that operate in both linear and Geiger modes. The quenching circuit is used to perform photon counting while minimising the afterpulsing phenomenon in the Geiger-mode APDs. The gain stabilization circuit is designed for the APD that operates in linear mode to control and maintain its gain independent of temperature and power supply fluctuations.

Bibliography

- [1] S.B. Alexander, Optical Communication Receiver Design, SPIE, 1997.
- [2] Aoife M. Moloney, "A CMOS Monolithically Integrated Photoreceiver Incorporating an Avalanche Photodiode", Ph.D. thesis, Department of Electrical and Electronic Engineering, University College Cork, April 2003.
- [3] D. Pellion, K. Jradi, A. Le Padellec, A. Rennane, F. Moutier, V. Borrel, D. Esteve, C. Magenc and A. R. Bazer-Bachi, "Geiger avalanche photodiodes as tentative light detectors for VHE gamma ray astronomy", Experimental Astronomy, Vol.27, No.3, pp.187-195, 2010.
- [4] K. Healy, B. Schiedt, and A. P. Morrison, "Solid-state nanopore technologies for nanopore-based DNA analysis", Future Medicine, Vol.2, No.6, pp.875-897, Dec. 2007.
- [5] A. Michael Krainak, X. Sun, G. Yang and W. Lu, "Comparison of linearmode avalanche photodiode lidar receivers for use at one-micron wavelength", Proc. SPIE 7681, 76810Y(2010)
- [6] F. Powolny, E. Auffray, H. Hillemanns, P. Jarron, P. Lecoq, T. C. Meyer and D. Moraes, "A Novel Time-Based Readout Scheme for a Combined PET-CT Detector Using APDs", IEEE Transactions on Nuclear Science, Vol.55, No.5, pp.2465-2474, Oct. 2008.

- [7] S. M. Sze. Semiconductor Devices Physics and Technology. Wiley, 2nd edition, 2001. ISBN:0471333727.
- [8] P. Bhattacharya. Semiconductor Optoelectronic Devices. Prentice Hall, 2nd edition, 1994. ISBN:0138057486.
- [9] P. A. Wolff, "Theory of electron multiplication in silicon and germanium", Physical Review, Vol.95, No.6, pp.1415-1420, Sep. 1954.
- [10] R. J. McIntyre, "A new look at impact ionization-Part I: A theory of gain, noise, breakdown probability, and frequency response", IEEE Transactions on Electron Devices, Vol.46, No.8, pp.1623-1631, Aug. 1999.
- [11] P. Yuan, K. A. Anselm, C. Hu, H. Nie, C. Lenox, A. L. Holmes, B. G. Streetman, J. C. Campbell, and R. J. McIntyre, "A new look at impact ionization-Part II: Gain and noise in short avalanche photodiodes", IEEE Transactions on Electron Devices, Vol.46, No.8, pp.1632-1639, Aug. 1999.
- [12] Donal Cronin, "Automatic Control Circuits for Geiger-Mode Avalanche Photodiodes", Ph.D. thesis, University College Cork, Ireland, Mar. 2009.
- [13] C. L. Anderson and C. R. Crowell, "Threshold Energies for Electron-Hole Pair Production by Impact-Ionization in Semiconductors", Physical Review B, Vol.5, No.6, pp.2267-2272, Mar. 1972.
- [14] X. Sun and F.M. Davidson, "Photon Counting with Silicon Avalanche Photodiodes", IEEE Journal of Lightwave Technol, Vol.10, No.8, pp.1023-1032, Aug. 2008.
- [15] H. Melchior and W.T. Lynch, "Signal and Noise Response of High Speed Germanium Avalanche Photodiodes", IEEE Transactions on Electron Device, Vol.13, No.12, pp.829-838, Dec. 1966.

- [16] Diarmuid Patrick O'Connell, "Control Circuits for Avalanche Photodiodes", Ph.D. thesis, University College Cork, Ireland, Feb. 2009.
- [17] R. B. Emmons, "Avalanche-photodiodes frequency response", Journal of Applied Physics, Vol.38, No.9, pp.3705-3714, Aug. 1967.
- [18] R. J. McIntyre, "Multiplication noise in uniform avalanche photodiodes", IEEE Transactions on Electron Devices, ED-13, pp.164-168, 1966.
- [19] S.M. Sze, Physics of Semiconductor Devices, John Wiley and Sons New York, 1981.
- [20] J. Kataoka, R. Saito, T. Ikagawa, J. Kotoku, Y. Kuramoto, Y. Tsubuku, T.Saito, Y. Yatsu, N. Kawai, Y. Ishikawa and N. Kawabata, "An active gain control system for Avalanche photo-diodes under moderate temperature variations", Nuclear Instruments and Methods in Physics Research A,Vol.564, pp.300-307, 2006.
- [21] J. Carlton Jackson, "Geiger-Mode Avalanche Photodiodes", Ph.D. thesis, National Microelectronics Research Centre, University College Cork, Ireland, Apr. 2003.
- [22] S. Cova, M. Ghioni, A. Lotito, I. Rech, and F. Zappa, "Evolution and prospects for single-photon avalanche diodes and quenching circuits", Journal of Modern Optics, Vol.51, No.9-10, pp.1267-1288, Jun./Jul. 2004.
- [23] A. Rochas, M. Gani, B. Furrer, P.A. Besse and R.S. Popovic, "Single photon detector fabricated in a complementary metal-oxide-semiconductor high-voltage technology", Review of Scientific Instruments, Vol.74, No.7, pp.3263-3270, Jul. 2003.
- [24] R.S. Muller and T.I. Kamins, "Device Electronics for Integrated Circuits", John Wiley and Sons Inc., second edition, 1986.

- [25] F. Zappa, M. Ghioni, S. Cova, A. Lotito and A. C. Giudice, "Monolithic Active- Quenching and Active-Reset Circuit for Single-Photon Avalanche Detectors", IEEE Journal of Solid-State Circuits, Vol.38, No.7, pp.1298-1301, Jul. 2003.
- [26] J. C. Jackson, A. P. Morrison, D. Phelan and A. Mathewson, "A Novel Silicon Geiger-Mode Avalanche Photodiode", Proceedings of International Electron Devices Meeting (IEDM), pp.797-800, Dec. 2002.
- [27] A. Rochas, M. Gani, B. Furrer, P.A. Besse and R.S. Popovic, "Single photon detector fabricated in a complementary metal-oxide-semiconductor high-voltage technology", Review of Scientific Instruments, Vol.74, No.7, pp.3263-3270, Jul. 2003.
- [28] Y. Kang, H. X. Lu, Y.-H. Lo, D. S. Bethune and W. P. Risk, "Dark count probability and quantum efficiency of avalanche photodiodes for singlephoton detection", Applied Physics Letters, Vol.83, No.14, pp.2955-2957, 2003.
- [29] M. Liu, X. Bai, X. Guo, J.C. Campbell, Z. Pan and M. M. Tashima, "Low Dark Count Rate and High Single-Photon Detection Efficiency Avalanche Photodiode in Geiger-Mode Operation", IEEE Photonics Technology Letters, Vol.19, No.6, pp.378-380, Mar. 2007.
- [30] F. Zappa, S. Tisa, A. Tosi and S. Cova, "Principles and features of singlephoton avalanche diode arrays", Sensor and Actuators A, Vol.140, pp.103-112, 2007.
- [31] S. Cova, M. Ghioni, A. Lacaita, C. Samori and F. Zappa, "Avalanche photodiodes and quenching circuits for single-photon detection", Journal of Applied Optics, Vol.35, No.12, pp.1956-1976, Apr. 1996.

- [32] V. Anna, A. Arbat, E. Vilella and A. Dieguez, "Geiger-Mode Avalanche Photodiodes in Standard CMOS Technologies", www.intechopen.com.
- [33] A.M. Moloney, A.P. Morrison, J.C. Jackson, A. Mathewson and P.J. Murphy, "Small signal equivalent circuit for geiger-mode avalanche photodiodes", IET Electronics Letters, Vol.38, No.6, pp.285-286, Mar. 2002.
- [34] http://www.ucc.ie/en/eleceng/research/researchareas/photodectect/
- [35] S.M. Sze and G. Gibbons, "Effect of junction curvature on breakdown voltage in semiconductors", Solid-State Electronics, Vol.9, No.9, pp.831-845, 1966.
- [36] J.C. Jackson, A.P. Morrison, P. Hurley, W.R. Harrell, D. Damjanovic, B. Lane and A. Mathewson, "Process monitoring and defect characterization of single photon avalanche diodes", Proceedings of the 2001 International Conference on Microelectronic Test Structures, pp.165-170, Mar. 2001.

Chapter 3

Biasing Avalanche photodiodes

It was shown in Chapter 2 that high bias voltages and mA range load currents are needed for the operation of APDs. This chapter presents a bias solution that meets these requirements and can be used for APDs operating in both linear and Geiger modes. In this chapter, the bias requirements for APDs and APD-based applications are introduced. A number of bias solutions are reviewed and discussed. The design and development of a dual-rail charge pump based bias circuit is described. The operation of this circuit is evaluated experimentally and its drawbacks identified. An improved bias circuit which simplifies the overall control of the bias voltage and eliminates the need for a negative voltage rail was developed and implemented. A layout of the circuit was designed using the L-Foundry 0.15 μ m process and post layout simulations are presented.

3.1 Introduction to APD Bias solutions

The bias voltage needed for the APD's operation is determined by its architecture. In this work, the APDs used are the planar shallow junction APDs developed by the Photodetection and Imaging Group at University College Cork [1]. These planar APDs operate with a much lower bias voltage (< 50 V) [2, 3] than that needed for Large Area APDs (1000 V ~ 2000 V) [4-6] and reach through APDs (300 V ~ 500 V) [7-9]. However, it's still difficult to bias these at voltages that are in excess of the voltage requirements for the rest of the sensing system which is typically 3.3 V or 5 V [10-13].

An external high-voltage supply such as that shown in Figure 3.1 is normally used to bias the APDs. This kind of supply is bulky, expensive and can not be used in portable devices.



Figure 3.1: Bench top power supply (Keithley 2400 digital sourcemeter)

Many step-up DC-DC converter chips are available in the market for biasing APDs. These chips are designed to have a high output voltage to bias the APD [14, 15]. However, several external discrete components are required for them to function correctly, as shown in Figure 3.2, thereby limiting their use-fulness in compact or integrated applications. The step-up DC-DC chips are mostly based on boost converters which are traditional high voltage generation circuits and widely used in discrete circuits [16, 17]. A basic schematic of a boost converter is shown in Figure 3.3.

The main drawback of these circuits is the use of inductors which lead to a large board area and height such as in [18]. When implementing an integrated circuit, the fabricated on-chip inductors are normally large and have low inductance and quality factors [19-21]. For example, the chip area required for







Figure 3.2: Commercial bias solutions for APDs: (a) MAXIM MAX5026, 5 V to 71 V APD power supply [14] and (b) LINEAR TECHNOLOGY LT3571, 5 V to 45 V APD bias power supply [15]



Figure 3.3: Basic schematic of a boost converter

a 58 nH inductor in [20] is 4 mm^2 and a 13.1 nH inductor in [21] occupied 0.8 mm^2 chip area. This drawback limits the boost converter's usefulness in on-chip step-up DC-DC converters which usually require large value inductors. Moreover, for most conventional CMOS processes, there is no standard inductor available in the libraries which makes designing an integrated boost converter with conventional CMOS processes difficult [22]. In [23], Yao-Yi Yang *et al.* presented a boost converter based bias circuit for APDs. The design can provide up to 80 V bias voltage for APDs with low output voltage ripple (after an external RC low pass filter). However, the use of a high value inductor (3.3 μ H) makes it incapable of being fully integrated.

Compared with boost converters, charge pump DC-DC converters are inductorless and the main components needed are capacitors and switches (diodes or transistors) which can be easily integrated on chip with currently available CMOS technologies. In [24, 25], charge pumps are used to provide on-chip power supply for biasing the APDs. With those solutions, the size, weight and power consumption of the APD-based systems are reduced. However, there are several drawbacks remain in those designs. Firstly, the bias voltages produced are not high (11 V in [24] and 25 V in [25]) that limit their usefulless for the APDs that require higher bias voltages (e.g. in this project the APD used has a breakdown voltage of 27 V and operates with the bias voltage from 25 V to 35 V). Moreover, in [24, 25], the maximum load current the bias circuits can provide are not mentioned which is a very important aspect for the bias circuits. If the maximum load current provided is small, when the illumination level to the APD is high or the APD arrays are used, the circuit will not be able to maintain the high bias voltage required. With the aim of developing an integrable bias circuit with high bias voltage and load current, as will be shown, a new regulated charge pumps are designed.

3.2 Design of an integrable bias circuit

For generating high output voltages and load current, regardless of the types of charge pumps, a large number of stages are required [26-29]. As the number of stages is increased, the threshold voltage of the MOS devices increases due to the body effect and the energy losses in each stage can significantly reduce the pumping gain and limit the circuit's performance [30]. With the same structure, fewer stages gives lower energy losses for each stage and less area occupation for the charge pumps. As will be shown, the dual-rail charge pump circuit proposed is utilized to generate a high potential between the positive and negative output with a small number of stages.

3.2.1 Circuit description

An overview of the proposed bias circuit is shown in Figure 3.4 and consists of two main parts: (i) A dual-rail charge pump which provides the high bias voltage for the load. (ii) Two shunt regulators which are used to control the output voltage for the positive and negative nodes of the load.



Figure 3.4: Block diagram of overall bias solution

Charge pumps

Figure 3.5 shows the schematic of the dual-rail charge pump. The positive and negative charge pumps used in Figure 3.5 are Dickson pump circuits [31]. The MOS transistors in the circuit are connected in diode fashion. The neighbouring pumping capacitors are connected to two inverse pumping clocks. NMOS is used for the positive charge pump to transfer the positive charge to the positive output. PMOS is used for the negative charge pump to transfer the positive charge away from the negative output [30].



Figure 3.5: Schematic of dual-rail charge pump



Figure 3.6: Positive charge pump

The charge pump shown in Figure 3.6 boosts the voltage at the positive output by the pumping clocks Clk and \overline{Clk} . At T1 when Clk is high and \overline{Clk} is low, diode D1 is forward biased and V1 is $(V_{dd} - V_d)$, where V_d is the forward voltage of the diode. Next, at time T2 when Clk is low and \overline{Clk} is high, V1 is pushed up to $(2V_{dd} - V_d)$. D2 is forward biased and V2 is $(2V_{dd} - 2V_d)$. At time T3 when Clk is high and \overline{Clk} is low, V1 is pushed up to $(3V_{dd} - 2V_d)$. In the same way, through every stage that consists of a capacitor-diode pair, the voltage is boosted stage by stage and at the output a high potential exceeding the power supply voltage can be obtained.



Figure 3.7: Negative charge pump

Similar to the positive charge pump, the negative charge pump is built on

capacitor-diode pairs. To achieve a negative output, the direction of the charge movement in the negative charge pump shown in Figure 3.6 is opposite to the positive charge pump. Through every stage, the positive charge is removed from the negative output that reduces its voltage from the initial level (here it is V_{dd}) to a negative level.

Shunt regulator

For the control of the charge pump's outputs, shunt regulators are used. With the shunt regulators, the charge pump operates continuously. Figure 3.8 shows the schematic of the shunt regulator. This regulator uses a MOS transistor (NMOS for positive output and PMOS for negative output) in parallel with the load and behaves functionally like a variable current divider to obtain load voltage regulation [32]. R1 and R2 are used to decrease the output voltage to a comparable level, V1.



Figure 3.8: Schematic of shunt regulator

A two-stage op-amp is used as the error amplifier (see Figure 3.9) [33]. The first stage is a differential input stage; the second stage is a common-source stage with an active load. The differential pair transistors have equal dimensions and are biased by a current mirror with one diode connected PMOS transistor as the load. All the transistors have a minimum length of 0.35 μ m. The widths of M1, M2, M3, and M4 are set to 25 μ m, M5 and M8 are set to 16 μ m, M6 is set to 15 μ m, M7 is set to 8 μ m and M9 is set to 2 μ m. A power supply of V_{dd} and GND are used for amplifying the positive error, a power supply of GND and $-V_{dd}$ is used for amplifying the negative error. Simulations show the DC Gain and GBW can reach 48dB and > 100MHz with a 20 pF load capacitor. The Phase Margin was simulated to 60 degrees that guarantees the stability of the amplifier (see Figure 3.10).



Figure 3.9: Schematic of two-stage op-amp



Figure 3.10: AC sweep of the 2-stage op-amp

With the error amplifiers, the operation of the shunt regulation shown in Figure 3.8 is described as follows: If V1 exceeds the regulation level V_{ref} , the error amplifier generates an output voltage proportional to the difference between V_{ref} and V1 and the MOS transistor will be biased to divide the output current. If the output falls below the regulation level, the MOS transistor will be turned off, and the output can be continuously charged up. In this way, V1 can be set equal to V_{ref} . By setting the reference voltage V_{ref} , the output voltage can be altered.

3.2.2 Simulations

The circuit was simulated using Austria Microsystem's 0.35 μ m CMOS process [34] running in the Cadence design environment. A 7-stage dual-rail charge pump with shunt regulators was used. The pumping capacitors are 20 pF. The two load capacitors are 3 nF each and are the only external components required for this design. The clock frequency is 50 MHz with $V_{dd} = V_{clk} = 5$ V.

Figure 3.11 demonstrates the regulation of the positive output voltage at 15 V. As can be seen from the figure, when the positive output reaches the regulation level, the error amplifier generates an output voltage to bias the NMOS transistor. The excess current to the output is divided by the NMOS transistor that maintains the positive output voltage equals to the regulation level.



Figure 3.11: Regulation of positive output to 15 V

Similar to the regulation of the positive output, the error amplifier in the negative charge pump circuit generates a negative voltage to bias the PMOS transistor. The excess negative current to the negative output is divided by the PMOS transistor that makes the negative output voltage equals to the regulation level (see Figure 3.12).



Figure 3.12: Regulation of negative output to -15 V

In Figure 3.13, Figure 3.14 and Figure 3.15, the bias voltages and the output voltages of the positive node and the negative node when the bias voltage is regulated at 25 V, 30 V and 35 V are illustrated. Simulation results show that the bias voltage can be regulated accurately with a peak-peak ripple of less than 25 mV.



Figure 3.13: Simulation results when the bias is regulated to 25 V



Figure 3.14: Simulation results when the bias is regulated to 30 V



Figure 3.15: Simulation results when the bias is regulated to 45 V

Figure 3.16 shows the maximum load current and output power for different bias voltages. From the simulations it is clear that the circuit is capable of meeting load current demands in excess of 1 mA at bias voltages up to 45 V.



Figure 3.16: Simulation results of maximum load current and output power for different bias voltages

3.2.3 Printed circuit board (PCB) implementation

With discrete components, the circuit was implemented on a printed circuit board (PCB). Due to large parasitics and speed limitations of a discrete implementation, larger capacitors are used. The pumping capacitors are 0.1 μ F and the load capacitors are 1 μ F. The N-transistors and P-transistors used are ZVN3306F and ZVP3306F, respectively. A TLE2021IP low-power precision op-amp is used as the error amplifier. The clock frequency is 100 kHz with V_{dd} = $V_{clk} = 5$ V.



Figure 3.17: Plots of maximum load current and output power for different bias voltages

In Figure 3.17, the maximum load current and output power for different bias voltages are demonstrated. It shows the circuit is capable of delivering greater than 1 mA at a bias voltage up to 40 V.



Figure 3.18: Oscilloscope trace when the bias voltage is regulated to 30 V.

Figure 3.18 shows the bias voltage when it is regulated at 30 V. The voltage ripples on the positive and negative output are 120 mV and 180 mV respectively; see Figure 3.19.



Figure 3.19: Voltage ripples on the positive and negative output.

This is mainly caused by two factors: (i) Noise due to the discrete implementation. (ii) Low speed of the amplifier, which means the amplifier cannot generate the error voltage quick enough when the output voltage exceeds the regulation level. The ripples can be reduced with an integrated circuit implementation or a faster amplifier [35, 36]. In summary, the circuit introduced is based on a dual-rail charge pump and was simulated in Cadence. A PCB module was fabricated and measured to verify the circuit function. Both simulations and experimental measurements show the circuit proposed is capable of providing in excess of 50 V for biasing planar APDs. Results also show that the output bias voltage can be adjusted precisely and can sustain mA output current up to 40 V bias voltage. However, there are several drawbacks to this circuit. In this design, two control voltages require adjusting by the end user to provide the required output voltage and a negative supply rail is needed to control the negative output, which adds to the complexity of the circuit. As will be seen in the next section, a new approach is designed to overcome these drawbacks.

3.3 New approach to the bias circuit

To overcome the limitations of the old design, a new approach was designed. In this circuit, pass-gate based voltage regulators replace the shunt regulators and the negative voltage regulator is now linked to the positive output, which links the bias voltage to the reference voltage of the positive voltage regulator. This simplifies the overall control of the bias voltage and eliminates the need for a negative voltage rail in the circuit. The layout of the circuit was designed using the L-Foundry 0.15 μ m process and the final design has a small footprint of 1.55 mm \times 1 mm.

3.3.1 Circuit description

Figure 3.20 shows the schematic of the new approach of the bias circuit which consists of: (i) Positive and negative charge pumps to provide the high bias voltage for the APD. (ii) Positive and negative voltage regulators that control the voltage of the positive and negative output.



Figure 3.20: Schematic of new bias circuit

The positive and negative charge pumps used are the same as described in Section 3.1. For the bias voltage regulation, pass-gates (PG1 and PG2) are connected between the input clocks and the charge pumps and controlled by the output of the operational amplifiers (Amp1 and Amp2). R1, R2 and Ra are potential dividers used to limit the input voltage range at nodes V1 and V2. V1 is connected to the non-inverting input (+) of Amp1 and V2 is connected to the inverting input (-) of Amp2.



Figure 3.21: Positive charge pump and regulation circuits

The regulation circuit for the positive output is shown in Figure 3.21. In the circuit, if V1 exceeds the regulation level, V_{ref} , the amplifier Amp1 generates an output voltage proportional to the difference between V_{ref} and V1. The pass-gate PG1 is turned off to block the input clocks and the positive output is decreased. If the output falls below the regulation level, the pass-gate will be turned on and the output can be charged up. In this way, V1 can be set equal to V_{ref} and the positive output voltage can be altered and V_{o+} can be calculated using

$$V_{o+} = V_{ref} \times \left(1 + \frac{R1}{R2}\right) \tag{3.1}$$



Figure 3.22: Negative charge pump and regulation circuits

The regulation circuit for the negative output is shown in Figure 3.22. In the circuit, if V2 is less than the set voltage, Vs, the amplifier Amp2 generates an output voltage proportional to the difference between Vs and V2. The pass-gate PG2 is turned off to block the input clocks to the negative charge pump, the amplitude of the negative output is decreased and V2 is increased. If V2 exceeds Vs, the pass-gate PG2 will be turned on, the amplitude of the negative output can be charged up and V2 will be decreased. In this way, V2 can be set equal to Vs and V_{o-} can be calculated using

$$V_{o-} = 2V_s - V_{o+} \tag{3.2}$$

The bias voltage V bias is the difference between V_{o+} and V_{o-}

$$V_{bias} = V_{o+} - V_{o-} \tag{3.3}$$

Replace V_{o+} and V_{o-} using (3.1) and (3.2), gives

$$V_{bias} = 2V_{ref} \times \left(1 + \frac{R1}{R2}\right) - 2V_s \tag{3.4}$$

In this circuit, the ratio of R1 and R2 is set to 9/1 and Vs is set to 1 V that gives

$$V_{bias} = 20 \times V_{ref} - 2 \tag{3.5}$$

As can be seen from (5), the bias voltage of the circuit can be regulated linearly by setting the reference voltage V_{ref} .
3.3.2 Layout and simulations

The layout was designed with L-foundry 0.15 μ m CMOS process [37]. A 5stage positive charge pump and a 6-stage negative charge pump are used. The pumping capacitors are set to 20 pF. Except for the 2 load capacitors, all the other components of the circuit are implemented on the chip.



Figure 3.23: Layout of the designed circuit

The op-amp schematic is shown in Figure 3.24(a) and the layout is shown in Figure 3.24(b). The pass-gate regulator schematic is shown in Figure 3.25(a) and the layout is shown in Figure 3.25(b). The completed circuit layout is shown in Figure 3.23 which has dimensions of 1.55mm \times 1mm. All the simulations reported are post-layout simulations.





Figure 3.24: (a) Schematic of the op-amp; (b) Layout of the op-amp.





Figure 3.25: (a) Schematic of the pass-gate regulator; (b) Layout of the pass-gate regulator.

For the simulations, the two load capacitors, set to 5 nF each, are the only external components of the circuit. The simulations were completed in the Cadence design environment. The clock frequency is 50 MHz with $V_{dd} = V_{clk}$ = 5 V.



Figure 3.26: Simulation results of maximum load current and output power for different bias voltages

Figure 3.26 shows the maximum load current and output power for different bias voltages. It is clear that the circuit is capable of meeting load current demands in excess of 1 mA for bias voltages up to 40 V.



Figure 3.27: Positive and negative output when the bias voltage is regulated to 30 V

Figure 3.27 shows the Positive and negative output when the bias voltage is regulated to 30 V. Here V_{ref} is set to 1.6 V that makes the positive output voltage equal to 16 V and the negative output voltage equal to -14 V (2 V -Positive output). As can be seen from the figure, with the voltage regulators, the bias voltage is precisely set to 30 V. With the accurate control of the bias voltage, the excess bias (bias voltage above the breakdown voltage of the APD) can be stabilized by using the "Time over Threshold" introduced in [25]. This method measures the dark photon count pulse width which is as a function of the excess bias, then adjusts the bias voltage to achieve a constant pulse width that makes the excess bias stable.



Figure 3.28: Bias voltages ripple and estimated chip size for different load capacitors

Figure 3.28 shows the bias voltage ripple for different load capacitors when the bias voltage is regulated at 30 V with a load current of 1 mA. The estimated chip size when the load capacitors are integrated on the chip is also illustrated. As can be seen from the figure, the ripple can be decreased to around 70 mV

when the values of the load capacitors are set to more than 500 pF. That means this circuit is capable of full integration if the bias voltage's ripple requirement of the devices is not very stringent (e.g. < 100 mV). [38]

3.4 Conclusions

This chapter has described the development of a bias circuit for avalanche photodiodes (APDs). The circuit design was explained and simulations were performed. A discrete version of the circuit was fabricated and tested. A layout of this circuit was also implemented with a conventional CMOS process. Results show that the circuit is capable of sourcing mA range load currents for shallow-junction planar APDs that operate up to 40 V. The layout of the circuit designed with the L-Foundry 0.15 μ m process shows a small footprint of $1.55 \text{ mm} \times 1 \text{ mm}$, making it suitable for hybrid integrated with the 2 external capacitors and an APD in a single package to operate seamlessly from a 5 V supply, thereby taking the additional responsibility for biasing the APD from the end-user. In the end, the possibility of developing a full integration version of this circuit was also discussed which shows the circuit is capable of fully integrated for the devices with voltage ripple tolerance of greater than 70 mV. Compared to the existing solutions that include external high-voltage supplies, DC-DC converter chips and reported designs [14, 15, 23], this bias solution is inductorless, highly integrable and provides a more compact solution that can be used to reduce size, weight and power consumption for APD-based applications.

Bibliography

- [1] http://www.ucc.ie/en/eleceng/research/researchareas/photodectect/
- [2] J. Jackson, J. Donnelly, B. O'Neill, A. Kelleher, G. Healy, A. Morrison and A. Mathewson, "A Integrated bulk/SOI APD sensor: Bulk substrate inspection with Geiger-mode avalanche photodiodes", IET Electronics letters, Vol.39, No.9, pp.735-736, 2003.
- [3] A. Moloney, A. Morrison, J. Jackson, A. Mathewson, J. Alderman, J. Donnelly, B. O'Neill, A. Kelleher, G. Healy and P. Murphy, "Monolithically integrated avalanche photodiode and transimpedance amplifier in a hybrid bulk/SOI CMOS process", IET Electronics letters, Vol.39, No.4, pp.391-392, 2003.
- [4] L. Shah, R. Farrell, R. Grazioso, R. Myers and L. Cirignano, "Largearea APDs and monolithic APD arrays", IEEE Transactions on Nuclear Science, Vol.48, No.6, pp.2352-2356, 2001.
- [5] R. Neilson, F. LePort, A. Pocar, K.S. Kumar, A. Odian, C.Y. Prescott, V. Tenev, N. Ackerman, D. Akimov, M. Auger, C. Benitez-Medina, M. Breidenbach, A. Burenkov, R. Conley, S. Cook, R. deVoe, M.J. Dolinski, W. Fairbank Jr., J. Farine, P. Fierlinger, B. Flatt, R. Gornea, G. Gratta, M. Green, C. Hall, K. Hall, D. Hallman, C. Hargrove, S. Herrin, J. Hodgson, L.J. Kaufman, A. Kovalenko, D.S. Leonard, D. Mackay, B. Mong, M.

Montero Diez, E. Niner, K. OSullivan, A. Piepke, P.C. Rowson, D. Sinclair, K. Skarpaas, S. Slutsky, V. Stekhanov, V. Strickland, K. Twelker, J.-L. Vuilleumier, K. Wamba, U. Wichoski, J. Wodin, L. Yang, Y.-R. Yen, "Characterization of large area APDs for the EXO-200 detector", Nuclear Instruments and Methods in Physics Research Section A: Accelerators, Spectrometers, Detectors and Associated Equipment, Vol.608, No.1, pp.68-75, Sep. 2009.

- [6] T.R. Gentile, C.D. Bass, J.S. Nico, H. Breuer, R. Farrell, "Magnetic field effects on large area avalanche photodiodes at cryogenic temperatures", Nuclear Instruments and Methods in Physics Research Section A: Accelerators, Spectrometers, Detectors and Associated Equipment, Vol.652, No.1, pp.520-523, Oct. 2011.
- [7] I. Corte ands, P. Ferne andndez-Marti andnez, D. Flores, S. Hidalgo and J. Rebollo, "Gain estimation of RT-APD devices by means of TCAD numerical simulations", 2011 Spanish Conference on Electron Devices (CDE), pp.1-4, 2011.
- [8] T. Nakamori, T. Enomoto, T. Toizumi, "Development of X-ray/gammaray imaging spectrometers using reach-through APD arrays", Journal of Instrumentation, 7(03): C03038, 2012.
- [9] S. Kou, Q. Chen, E. Ji, W. Xu and Jian Qin, "Single-photon time response of reach-through APD", Optics Communications, Vol.281, No. 9, pp.2481-2484, May 2008.
- [10] J. Macri, A. Wintenberg, M. Widholm, U. Jagadish, J. Ledoux, M. Mc-Connell, S. Frank and H. Cutlip, "Readout of scintillating plastic fibers with an APD array and prototype ASIC", IEEE Transactions on Nuclear Science, Vol.50, No.4, pp.928-935, 2003.

- [11] A. Nassalski, M. Moszyn andski, A. Syntfeld-Kaz anduch, T. Szcze ands andniak, L. S andwiderski, D. Wolski, T. Batsch and J. Baszak, "Multi Pixel Photon Counters (MPPC) as an Alternative to APD in PET Applications", IEEE Transactions on Nuclear Science, Vol.57, No.3, pp.1008-1014, 2010.
- [12] E. Albuquerque, V. Bexiga, R. Bugalho, B. Carrio, C.S. Ferreira, M. Ferreira, J. Godinho, F. Gonalves, C. Leong, P. Lous, P. Machado, R. Moura, P. Neves, C. Ortigo, F. Piedade, J.F. Pinheiro, J. Rego, A. Rivetti, P. Rodrigues, J.C. Silva, M.M. Silva, I.C. Teixeira, J.P. Teixeira, A. Trindade, J. Varela, "Experimental characterization of the 192 channel Clear-PEM frontend ASIC coupled to a multi-pixel APD readout of LYSO:Ce crystals", Nuclear Instruments and Methods in Physics Research Section A: Accelerators, Spectrometers, Detectors and Associated Equipment, Vol.598, No.3, pp.802-814, Jan. 2009.
- [13] M. Koizumi, J. Kataoka, S. Tanaka, H. Ishibashi, N. Kawai, H. Ikeda, Y. Ishikawa, N. Kawabata, Y. Matsunaga, K. Shimizu and H. Kubo, "Development of a low-noise analog front-end ASIC for APD-PET detectors", Nuclear Instruments and Methods in Physics Research Section A: Accelerators, Spectrometers, Detectors and Associated Equipment, Vol.604, No.1-2, pp.327-330, Jun. 2009.
- [14] http://www.maxim-ic.com/app-notes/index.mvp/id/1831
- [15] http://cds.linear.com/docs/Datasheet/3571fa.pdf
- [16] Y. Berkovich and B. Axelrod, "Switched-coupled inductor cell for DC-DC converters with very large conversion ratio", IET Power Electronics, Vol.4, No.3, pp.309-315, 2011.

- [17] G. Henn, R. Silva, P. Pra anda, L. Barreto and D. Oliveira, "Interleaved-Boost Converter With High Voltage Gain", IEEE Transactions on Power Electronics, Vol.25, No.11, pp.2753-2761, 2010.
- [18] K. I. Hwu and Y. T. Yau, "Voltage-Boosting Converter Based on Charge Pump and Coupling Inductor with Passive Voltage Clamping", IEEE Transactions on Industrial Electronics, Vol.57, No.5, pp.1719–1727, May 2010.
- [19] D. S. Gardner, G. Schrom, F. Paillet, B. Jamieson, T. Karnik, and S. Borkar, "Review of on-chip inductor structures with magnetic films", IEEE Transactions on Magnetics, Vol.45, No.10, pp.4760-4766, Oct. 2009.
- [20] C. Martin, B. Allard, D. Tournier, M. Soueidan, J.-J Rousseau, D. Allessem, L. Menager, V. Bley and J-Y. Lembeye, "Planar inductors for high frequency DC-DC converters using microwave magnetic material", IEEE Energy Conversion Congress and Exposition (ECCE 2009), pp.1890-1894, Sep. 2009.
- [21] R. Wu and J.K.O. Sin, "A Novel Silicon-Embedded Coreless Inductor for High-Frequency Power Management Applications", IEEE Electron Device Letters, Vol.32, No.1, pp.60-62, Jan. 2011.
- [22] http://www.europractice-ic.com/technologies.php
- [23] Yao-Yi Yang, Chun-Yu Hsieh, Tzu-Chi Huang, Yu-Huei Lee, Shin-Wei Wang, Ming-Yan Fan, Ming-Jhe Du, Shih-Hsien Cheng and Ke-Horng Chen, "A 80V output voltage boost converter with low voltage ripple for Avalanche Photodiode(APD)", IEEE International Symposium on Circuits and Systems (ISCAS), pp.757-760, May 2011.
- [24] R.K. Henderson, E.A.G. Webster, R. Walker, J.A. Richardson and L.A. Grant, "A 3×3, 5μm pitch, 3-transistor single photon avalanche diode

array with integrated 11V bias generation in 90nm CMOS technology", 2010 IEEE Internationa Electron Devices Meeting (IEDM), pp.14.2.1-14.2.4, Dec 2010.

- [25] S. Mandai and E. Charbon, "A 3.3-to-25V all-digital charge pump based system with temperature and load compensation for avalanche photodiode cameras with fixed sensitivity", Journal of Instrumentation, Vol.8, no.3, pp.030131-0301310, Apr. 2013.
- [26] M.R. Hoque, T. Ahmad, T. McNutt, A. Mantooth, M.M. Mojarradi, "Design technique of an on-chip, high-voltage charge pump in SOI", IEEE International Symposium on Circuits and Systems (ISCAS), pp.133-136, May 2005.
- [27] Ming-Dou Ker and Shih-Lun Chen, "Ultra-High-Voltage Charge Pump Circuit in Low-Voltage Bulk CMOS Processes With Polysilicon Diodes", IEEE Transactions on Circuits and Systems II: Express Briefs, Vol.54, No.1, pp.47-51, Jan. 2007.
- [28] J. Ding; Y. Wang; S. Jia, G. Du and X. Zhang, "Clock swing enhanced charge pump", International Conference of Electron Devices and Solid-State Circuits (EDSSC), pp.1-2, Nov. 2011.
- [29] L. Shen and K. Hofmann, "Fully integratable 4-phase charge pump architecture for high voltage applications", Proceedings of the 19th International Conference Mixed Design of Integrated Circuits and Systems (MIXDES), pp.265 -268, May 2012.
- [30] F. Pan and T. Samaddar, Charge pump circuit design, Recherche, 2006.
- [31] J. Dickson, "On-chip high-voltage generation in MNOS integrated circuits using an improved voltage multiplier technique", IEEE Journal of Solid-State Circuits, Vol.11, No.3, pp.374-378, 1976.

- [32] J.J. Chen, M.S. Lin, C.M. Kung and Y.S. Hwang, "Low-quiescent-current fast-response current-feedback shunt regulator", IEEE Asia Pacific Conference on Circuits and Systems (APCCAS2008), pp.530-533, 2008.
- [33] P.E. Allen and D.R. Holberg, CMOS analog circuit design, Oxford, 2002.
- [34] http://asic.austriamicrosystems.com/
- [35] S. Deng and A.P. Morrison, "Integrable bias solution for avalanche photodiodes", IET Electronics letters, Vol.47, No.4, pp.240-241, Feb. 2011.
- [36] S. Deng and A.P. Morrison, "A Dual-rail Charge Pump Bias Circuit for Avalanche Photodiodes", 22nd IET Irish Signals and Systems Conference (ISSC), pp.56-58, Jun. 2011.
- [37] http://www.lfoundry.com/
- [38] S. Deng and A.P. Morrison, "Design of an Adjustable Bias Circuit Using a Single-Sided CMOS Supply for Avalanche Photodiodes", 10th IEEE International NEWCAS Conference (NEWCAS2012), pp.65-68, Jun. 2012.

Chapter 4

Active quench and reset circuit (AQRC) for Geiger-mode avalanche photodiodes

As described in Chapter 2, the afterpulsing phenomenon is an unwanted source of noise for the GM-APD which needs to be minimized. This is accomplished by properly setting the hold-off time using a quenching circuit. This chapter introduces the design and development of a new active quench and reset circuit (AQRC) for Geiger-mode avalanche photodiodes (GM-APDs). Firstly, several previously reported quenching circuits and hold-off time control techniques are reviewed. The merits and drawbacks for each circuit are discussed. Next, a new AQRC with high-resolution hold-off time control logic is designed to overcome the drawbacks in the existing AQRCs. A layout of the circuit is designed and post-layout simulations are performed to evaluate the operation of the circuit. Following that, an improved approach is presented that simplifies the end-user control. The circuit was fabricated using AMS 0.35 μ m CMOS process and tested with a GM-APD.

4.1 Quenching circuits and hold-off control techniques

In APD-based photon counting applications, the photo current generated in the GM-APDs must be eliminated after an avalanche event. This is accomplished by using passive or active quenching circuitry. A passive quenching circuit is the simplest way to quench the avalanche in a GM-APD. In passive quenching, the GM-APD is connected in series with a high value resistor. The operation of a passive quenching circuit is illustrated in Figure 4.1.



Figure 4.1: Passive quenching circuit

When an avalanche event happens in the GM-APD, a large voltage increases by the large series resistor at the GM-APDs anode and the bias across the GM-APD is quenched to its breakdown voltage. During this process, the diode current rises to its peak value which determined by the values of excess bias voltage and the series resistor. The quenching time can be defined as:

$$t_q \cong (C_j + C_p) R_d \tag{4.1}$$

Where C_j is the junction capacitance of the APD, Cp is the parasitic capacitance at the anode of the APD and Rd is the equivalent resistance in the APD. Typically, the APD is connected in series with an external resistor and the parasitic capacitance C_p can be from several pF up to tens of pF which results in a quenching time of > 100 ns as reported in [1], [2] and [3]. Many passive quenching circuits reported integrate the series connected external resistor directly with the APD that makes the parasitic capacitance much smaller to reduce the quenching time [4-6]. After the GM-APD has been quenched, the bias voltage slowly reset to its original voltage with a RC time constant determined by the parasitic capacitance and the series connected resistor. For most passive quenching circuits reported in the literature, such as in [7-11], the reset time is typically in the microsecond region. The reset time can be reduced to tens of nanoseconds, but it requires integration of the APD and quenching circuit together to decrease the parasitics [4-6]. In addition to the long quench and reset times, there is another disadvantage of using the passive quenching circuit. During the reset process, the bias voltage is above the breakdown voltage and the avalanche events can be triggered before the bias returns to the original voltage. When the counting rate is high, this will significantly affect the APDs performance such as detection efficiency, timing response, etc.

In summary, the passive quenching circuit is simple, low cost and occupies a small area that makes it useful in applications such as the fabrication of large APD arrays such as in [12-17]. However, it has the drawback of very slow quench and reset times (in the microsecond range) that limit the photon counting rate. Moreover, the ill-defined dead time leads to the possibility of avalanche retriggering and thus increased afterpulsing.

To overcome these drawbacks, active quench and reset circuits (AQRCs) are used. The AQRC quenches the avalanche current by lowering its bias voltage below the breakdown voltage after an avalanche event. The GM-APD is then kept in the OFF state for fixed period of time, called the hold-off time, before resetting the device to its original bias voltage to await the next avalanche event. With the AQRC, the dead time of the GM-APD can be controlled to effectively reduce the afterpulsing phenomenon. Moreover, the stable well defined hold-off time (dead-time) make it possible to apply dead time correction to account for photons absorbed during hold-off time when an avalanche can't be triggered [18].



Figure 4.2: Typical structure of an active quench and reset circuit

Figure 4.2 shows the typical structure of an AQRC. The sensing circuit is used to sense the avalanche current in the GM-APD. This is typically a comparator or a simple inverter. The hold-off time control circuit receives a signal from the sensing circuit and generates pulses to control the quench and reset switches for holding the GM-APD in the OFF state for a period of time and then resetting it. The first AQRC was introduced in 1975 and was built using discrete components to enhance the performance of GM-APDs [19]. This circuit is designed with the aim of shortening the bias voltage reset time and providing a well defined hold-off time (dead time). In 1996, a compact discrete AQRC module was developed [20] which employed monostables to set the hold-off time. Progress in AQRCs made by the same research group led to hybrid/integrated versions of the AQRC [21, 22], enabling the possibility of developing miniaturized detector modules.



Figure 4.3: Diagram of AQRC in [21]

The AQRC [21] shown in Figure 4.3 consists of: 1) a comparator that senses the avalanche event; 2) monostables that generate the pulses for quenching and resetting the APD. As shown in Figure 4.3, the GM-APD is biased above its breakdown voltage. The excess bias voltage is about 10% of the breakdown voltage. When an avalanche event occurs a voltage is generated across the resistor Rs and sensed by a fast comparator. The comparator is used to provide an output pulse to external counters and to trigger a monostable multivibrator, $M_{hold-off}$.

The schematic of the monostable is shown in Figure 4.4. An RC delay circuit is used to generate a pulse for quenching or resetting the APD. With the monostable, the hold-off time can be adjusted by setting the value of the external resistor. As soon as the monostable is triggered, the QUENCH goes low, the quench switch (S_Q) is closed and this increases the anode voltage to Vquench. Vquench is higher than the excess bias voltage and thus brings the APD bias



Figure 4.4: Schematic of a monostable

below its breakdown voltage. During the hold-off time period, the GM-APD remains quenched and no avalanche can be detected. When $M_{hold-off}$ resets (QUENCH goes high), the quench switch S_Q is opened and the monostable M_{reset} is triggered. The signal RESET goes high to close the switch S_R and this lowers the anode potential back to ground. After the transition of the M_{reset} monostable, the switch S_R is opened and the GM-APD is ready to detect another photon. This circuit achieved a dead-time of 40 ns between consecutive avalanche breakdowns (giving a maximum count-rate of 25 Mcounts/s) and shows the capability of providing a user-controllable hold-off time to reduce afterpulsing.



Figure 4.5: Modified monostable proposed in [23]

A modified monostable was proposed in [23], see Figure 4.5. In this circuit,

a WIDTH transistor is used to replace the resistor. The hold-off or reset monostable pulse widths can be adjusted by varying the input voltage of the WIDTH transistor. As reported in [23], this design is capable of setting the hold-off time between 4.6 μ s and 4 ns by varying the input voltage between 0.635 V and 2.5 V. This makes it easier to adjust the hold-off time for an AQRC. Up to now, the use of monostables remains the most popular method for setting the hold-off time [24-27] with circuits still using this technique reported recently [28].



Figure 4.6: Schematic of AQRC reported in [29] and [30].

Figure 4.6 shows another AQRC reported by Rochas *et al* in which the GM-APD and the control circuit are integrated together. This circuit was implemented in both discrete and integrated formats reported in [29] and [30] respectively. Similar to the design in [21], this circuit uses a comparator to sense the avalanche events and a monostable multivibrator for controlling the hold-off time. Thanks to the monolithic integration, the parasitics of this circuit are greatly reduced and a dead-time of just 10 ns between adjacent events was achieved that leads to a very high maximum count-rate of 100 Mcounts/s. Some other designs that integrate the quenching circuit and the APD together

can be seen in [31, 32] in which maximum count-rates of excess 150 Mcounts/s are achieved. The monolithic integration gives a high maximum count-rate but it makes it difficult for users to interchange the sensor.

One drawback that remains in the AQRCs described is the hold-off time setting with monostables. With monostables, the adjustment of the hold-off time is non-linear and it's difficult to select the optimal hold-off time digitally or via additional signal processing circuitry. To overcome this drawback, the delay line technique can be used to set the hold-off time in AQRC. Figure 4.7 shows the schematic of an AQRC using a delay line technique to set the hold-off time proposed in [33].



Figure 4.7: Diagram of AQRC in [33]

In this circuit, a comparator is used to sense the avalanche event in the GM-APD and send signals to a pulse generator. The pulse generators are based on a silicon delay line with additional combinational logic that is used to create pre-defined pulse-widths for setting the hold-off time.



Figure 4.8: Diagram of delay line technique in [33]

A diagram of the delay line technique used in [33] is shown in Figure 4.8. The rising-edge ramp generator in the figure is a current source connected through a capacitor to ground. A current source/current mirror configuration is used to create the rising ramp voltage V_{x1} when the gate voltage, d1, of the NMOS transistor W_{n1} is set low. 16 discrete current mirrors were used with each having an individual capacitor to generate 16 predefined delays. The delay can be selected using a 4-16 decoder via inputs h3 to h0. The outputs of the 16 SR-latches are connected to a 16-1 multiplexor and the same inputs h3 to h0 selected for the decoder are used to determine the SR-latch output sent to the next stage. This output is then connected to a logic stage to create one of

h3 h2 h1 h0	Hold-off time	h3 h2 h1 h0	Hold-off time
0000	9.5 ns	1000	276 ns
0001	17.5 ns	1001	317 ns
0010	42.5 ns	1010	363 ns
0011	71 ns	1011	375 ns
0100	101 ns	1100	444 ns
0101	142 ns	1101	486 ns
0110	187 ns	1110	588 ns
0111	224 ns	1111	683 ns

Table 4.1: Available hold-off times with AQRC in [33]

the selected 16 predefined high hold-off time pulse widths between 5 and 660 ns, ultimately determined by the size of the capacitors C1-C16 and the current flowing through them. After the hold-off time is completed, the active quench stage is deactivated (parallel NMOS transistors of Figure 4.7) and the parallel PMOS transistors are turned on using a logic low pulse. This is provided by a similar pulse generator operating in 16 discrete steps between 2.5 and 100 ns.

4.2 High-resolution hold-off time control circuit

As discussed in Section 4.1, the adjustment of the hold-off time in previously reported quenching circuits are either non-linear or complicated. To overcome the drawbacks, a simplified high-resolution hold-off time circuit is designed.

4.2.1 Circuit Design

Figure 4.9 shows the block diagram of the proposed circuit. The non-inverting input of the comparator is connected to the anode of the APD, which is biased at the voltage between the avalanche breakdown voltage, V_{break} , and $(V_{break} + V_{dd})$. The comparator is used to sense the avalanche current at the

anode of the APD which also has an inverse output, *compo*, that is connected to an external bond pad for readout. One PMOS and one NMOS transistor are used as the switches for quenching and resetting the APD. An external clock signal provides the counting clock during the hold-off period and a counter is used to control the hold-off time.



Figure 4.9: Block diagram of high-resolution hold-off time control circuit

Initially, when there is no avalanche current, compo is low, the external clock is blocked, the 6-bit counter is reset to 0 ("000000") and both PMOS and NMOS transistors are turned off. When an avalanche event happens in the APD, current flows through the load resistor, R_L , and the voltage increases at the anode of APD. The comparator senses the voltage rise and compo goes from low to high. Qp goes low to turn on the PMOS transistor and the anode of the APD is connected to V_{dd} for quenching. Meanwhile, the counter receives clocks from the external clock, Clk(in).

The counter used here is a 6-bit synchronous binary counter which consists of 6 J-K flip-flops with the clock signal connected to the clock input of every flip-flop with the J and K inputs tied together; see Figure 4.10. The J and K inputs of the first flip-flop are connected to V_{dd} ; the J and K inputs of the other flip-flops are connected to the output, Q, of each front end. When the



Figure 4.10: 6-bit synchronous binary counter

reset signal compo is high, the counter receives clocks from Clk(in) and counts upwards from 0 ("000000") to 63 ("111111"). Each output of the counter is connected to one input of an XNOR gate. The other input of the XNOR is connected to an external input (controlled by the end user). When the output of the counter is equal to the external inputs, all the outputs of the XNOR gates go to logic "1" (high). Then Rn goes high which makes Qp go high to stop the hold-off process and turn on the NMOS transistor to reset the APD (two buffers are used here to make sure the reset process starts after the holdoff process is finished). At this time, the Node A' goes low to stop the clock to the counter and the counter is stopped. This then makes Rn remain high for resetting. When the anode of the APD is reset back to ground, compo is low, the Clk(in) is blocked again and the counter is reset to 0 ("000000"). Now the outputs of the counter do not match the external inputs, Rn goes low and the NMOS transistor is turned off to complete the reset process. The APD is then ready to detect the next photon. By setting the external inputs, the counting number can be determined and the hold-off time can be altered. The step resolution is decided by the counting speed, which depends on the period of the external clock Clk(in).

4.2.2 Layout and simulations

The layout of the proposed circuit was completed using L-Foundry 0.15 μ m CMOS process [34] and is illustrated in Figure 4.11. The overall chip dimension is about 700 μ m × 700 μ m which is mostly occupied by the bond pads and the I/O standard cells. The dimensions of the IC core (without bond pads) are 95 μ m × 55 μ m. All the simulations reported are post-layout simulations.



Figure 4.11: Layout of the proposed circuit

For circuit simulations, a linear model of the GM-APD is used, as illustrated in Figure 4.12 [20]. V_b is a voltage source that represents the breakdown voltage, which is set at 27 V. The bias voltage is set to 30 V. R_d is the internal resistance, which is set to 250 Ω . C_d is the junction capacitance, which is set to 2 pF. The simulations were run in the Cadence design environment with $V_{dd}=3.3$ V.



Figure 4.12: Simulation model of the GM-APD

Figure 4.13 shows an example of the circuit operation when the external inputs are set to 30 ("011110"). The period of the external clocks is set to 2 ns. As can be seen from the figure, at 5 ns when a photon absorbed, the comparator senses the voltage change from the anode of the APD and Qp goes low for quenching. Meanwhile, the external clock is not blocked and provides the clock pulses to the counter. The 6-bit counter counts upwards from 0 ("000000") at a rate set by the external clocks (here the period is set to 2 ns). When the outputs of the counter match the external inputs, which in this case are set to 30 ("011110"), Rn goes high which makes Qp go high to stop the hold-off process and turn on the NMOS transistor to reset the APD. At this time, the clocks to the counter are blocked and the counter is stopped thereby keeping Rn high for resetting the APD. When the anode of the APD is set back to ground, the counter is reset to 0 ("000000") and Rn goes low to stop the reset process. In this way, with the external inputs of 30 ("011110"), the hold-off time is set to around 60 ns.



Figure 4.13: Example of the circuit operation when the external inputs are set to 30 ("011110")

Figure 4.14 shows the simulation results of varying the external input codes versus the resultant hold-off time. It shows when the input code increases from 1 ("000001") to 63 ("111111") the hold-off time linearly increases to more than 120 ns with a step resolution of about 2 ns.



Figure 4.14: Hold-off times versus external input codes when the step resolution is set to 2 ns

Figure 4.15 shows the setting range of the hold-off time for different step resolutions. As can be seen from the figure, when the step resolution is varied from 2 ns to 20 ns, the range of the hold-off can be altered by more than a microsecond.



Figure 4.15: Setting range of the hold-off time for different step resolutions

In summary, this section has introduced a high-resolution hold-off time control circuit for GM-APDs. Circuit design, layout and simulations were presented. With this circuit, the hold-off time can be linearly varied from several nanoseconds to microseconds with a user-set resolution. The optimal 'afterpulse-free' hold-off time for any GM-APD can be easily set through the circuit's digital inputs or via an additional signal processing circuit. The layout of this circuit was completed using a conventional CMOS process, resulting in a small layout area that makes it suitable for integration with arrays of GM-APDs. The circuit also incorporates a facility designed to reset the APD automatically at the end of the hold-off time that further simplifies the control for the end-user. However, the use of the external clock is replaced by a ring-oscillator. With that configuration, the clock pulses to the counter can be generated locally and this simplifies the end-user control. [35].

4.3 Improved approach and ASIC development of the active quench and reset circuit

To simplify the user control of the design described in Section 4.2, an improved circuit was developed. In this new approach, a ring-oscillator replaces the external clock to simplify the end-user control. This circuit was fabricated using Austria Microsystems (AMS) 0.35 μ m CMOS process and tested with a planar silicon GM-APD having a 27 V breakdown voltage.

4.3.1 Circuit description

Figure 4.16 shows the block diagram of the proposed circuit. The inverting input of the comparator is connected to the cathode of the APD which is biased at a voltage determined by V_{dd} and $-V_{low}$. A comparator is used to sense the avalanche current at the cathode of the APD. One PMOS and one NMOS transistor are used as the switches for quenching and resetting the APD. A ring-oscillator is used to generate clock pulses during the hold-off period and a counter is used to count the clock pulses to adjust the hold-off time.



Figure 4.16: Block diagram of the new approach

When there is no avalanche current, compo is low, the ring-oscillator is inactive, the 8-bit counter is reset to 0 ("00000000") and the PMOS and NMOS transistors are turned off. When an avalanche event occurs, current flows through the sensing resistor Rs and a voltage drop is observed at the cathode of the APD. The comparator senses the voltage drop and compo goes from low to high. Qp goes high to turn on the NMOS transistor and the cathode of the APD is connected to GND for quenching. Meanwhile, the ring oscillator is active and provides the clock to the counter. The counter used is an 8-bit synchronous binary counter which has the same architecture as that was described in Section 4.2.1. When the reset signal compo is high, the counter receives clocks from the ring-oscillator and counts upwards from 0 ("00000000") to 255 ("11111111"). Each output of the counter is connected to one input of an XNOR gate. The other input of the XNOR is connected to an external input (controlled by end user). When the outputs of the counter equal the external inputs, all the outputs of the XNOR gates go to logic "1" (high). Then Rn goes low which makes Qp go low to stop the hold-off process and turn on the PMOS transistor thereby resetting the APD (buffers are used here to make sure the reset process starts after the hold-off process is finished). At this time, Node A goes low to block the clock from the ring-oscillator to the counter and the counter is stopped. This makes Rn remain low for resetting. When the cathode of the APD is reset back to V_{dd} , compo is low, the ring-oscillator is inactive and the counter is reset to 0 ("00000000"). Now the outputs of the counter do not match the external inputs, Rn goes high and the PMOS transistor is turned off to complete the reset process. The APD is then ready to detect the next photon. By setting the external inputs, the counting number can be determined and the hold-off time can be altered. The hold-off time setting step is determined by the counting speed that depends on the number of the stages of the ring-oscillator. A 41-stage ring-oscillator occupying approximately 25% of the IC core is used here to give a stable setting step of 6.5 ns for the hold-off time.

4.3.2 ASIC development and measurements

The circuit layout was designed with 0.35 μ m AMS CMOS process [36] and submitted to Europractice-IC [37] for fabrication. Figure 4.17 shows a photograph of the fabricated chip; the overall chip dimensions are 1.7 mm × 1.4 mm which mostly occupied by the bond pads and decoupling capacitors. The dimensions of the IC core are 260 μ m × 150 μ m.



Figure 4.17: Photograph of the fabricated chip

For the measurements, the AQR-IC is connected to a 20 μ m diameter circular GM-APD developed by the Photodetection and Imaging Group at University College Cork [38-40]. The fabricated PCB can be seen in Figures 4.18 and 4.19.



Figure 4.18: Top-side of the test PCB (holds the fabricated chip)



Figure 4.19: Bottom-side of the test PCB (holds the tested APDs)

In Figure 4.20, the experimental setup for the hold-off time test is demonstrated. A He-Ne laser source operating at 632.8 nm was used as the light source that was directed to the GM-APD using an integrating sphere. An external power supply was used to provide $-V_{low}$ and V_{dd} for biasing the GM-APD and powering the AQR-IC. The APD is biased at 30 V ($-V_{low} = -26.7$ V, $V_{dd} =$ 3.3 V) which is around 3 V in excess of the breakdown voltage. The hold-off time is controlled by the 8 way switch on the PCB. An oscilloscope was used to observe the quench pulse (Qp) and the voltage at the cathode ($V_{cathode}$) of the APD.



Figure 4.20: Experimental setup for testing the AQR-IC

In Figure 4.21(a), the oscilloscope traces of the quenching pulse (Qp) and the cathode voltage ($V_{cathode}$) are demonstrated with the hold-off time kept at 190 ns by setting Input7 to Input0 to 00011101. Figure 4.21(b) and Figure 4.21(c) show the cathode voltage ($V_{cathode}$) of the APD with the hold-off times set to 326 ns and 1.18 μ s. The results demonstrate the accurate control of the hold-off time using external switches.



Figure 4.21: a) Quenching pulse (Qp) and APD's cathode voltage ($V_{cathode}$) with a hold-off time of 190 ns (00011101), (b) $V_{cathode}$ with hold-off time = 326 ns (00110010) and (c) $V_{cathode}$ with hold-off time = 1.18 μ s (10110101).

Figure 4.22 shows the results of varying the external input codes versus the resultant hold-off time. It shows when the input code increases from 1 ("00000001") to 255 ("11111111") the hold-off time linearly increases from several nanoseconds to more than 1.6 μ s with a setting step of about 6.5 ns.



Figure 4.22: External input codes versus hold-off time

In Figure 4.23, a plot of $V_{cathode}$ is shown with Input7 to Input0 set to "00000001" when saturating light is directed at the APD; this demonstrates the minimum dead-time between adjacent avalanche events in the GM-APD. It can be seen that the minimum dead-time is 28.4 ns corresponding to a saturated count-rate of 35.2 Mcounts/s with this AQR-IC.[41, 42].



Figure 4.23: Plot of $V_{cathode}$ for saturated count-rate

4.4 Conclusion

This chapter described the design and development of an active quench and reset circuit with high-resolution hold-off time control logic. The circuit operation was explained, layout design and simulations were presented. The integrated circuit was fabricated using a conventional CMOS process and tested with a GM-APD.

Results show that this design is capable of linear changes to the hold-off time from several nanoseconds to microseconds with a reasonable step of 6.5 ns that allows setting of the optimal 'afterpulse-free' hold-off time for any GM-APD through digital inputs or additional signal processing circuitry. A minimum dead time of 28.4 ns was observed from the measurement, demonstrating a saturated photon-counting of 35.2 Mcounts/s with this AQRC.

A comparison of existing active quench and reset circuits and the circuit designed in this work is shown in Table 4.2. Monostables were used by many research groups to select the hold-off time and reset pulse. The main disadvantage of using monostables is the difficulty in accurately adjusting the hold-off time. The monostable adjusts the hold-off time by setting the value of an external resistor or varying the input voltage of a WIDTH transistor that leads to a non-linear change in the hold-off time, thereby making it difficult to select the optimal hold-off time digitally. Delay line techniques provide a more advanced way to set the hold-off time. The digitally controlled hold-off time makes it easier to interface with other signal processing circuitry. However, there are limited hold-off times available with the circuit and an increased chip areas needed if additional hold-off times are required. The circuit designed in this work allows digital control of the hold-off time while maintaining a comparable maximum count-rate to previous designs (as presented in [28,29], monolithic
integration of the APD and the quench circuit would produce a much higher saturated count-rate). Moreover, it eliminates the requirement of using an additional monostable or pulse generator to reset the detector, thus simplifying the circuit.

	[2]	[21, 22]	[31, 32]	[33]	This work
Integrated	No	Yes	Yes	Yes	Yes
Connection to the	not integrated	not integrated	Monolithic integrated	PCB	PCB
APD					
Min. Deadtime	39 ns	50 ns	5.4 ns	40 ns	28.4 ns
Max. Count-rate	25.6	20	185	25	35.2
(Mcounts/s)					
Hold-off time con-	Monostable	Monostable	Monostable	Delay line	Novel control logic
trol logic type					
Hold-off time ad-	External resistor	External resistor	Fixed	Digitally	Digitally
justment					
Additional control	Yes	Yes	Yes	Yes	No
needed for resetting					
APD					

Table 4.2: Comparison of existing active quench and reset circuits

Bibliography

- C. Niclass, M. Gersbach, R. Henderson, L. Grant, and E. Charbon, "A single photon avalanche diode implemented in 130-nm CMOS technology", IEEE Journal of Selected Topics in Quantum Electronics, Vol.13, No.4, pp.863-869, Jul./Aug. 2007.
- [2] A. D. Mora, A. Tosi, S. Tisa, and F. Zappa, "Single-photon avalanche diode model for circuit simulations", IEEE Photonics Technology Letters, Vol.19, pp.1922-1924, Dec. 2007.
- [3] Donal Cronin, "Automatic Control Circuits for Geiger-Mode Avalanche Photodiodes", Ph.D. thesis, Department of Electrical and Electronic Engineering, University College Cork, Mar. 2009.
- [4] C. Niclass, A. Rochas, P.-A. Besse, and E. Charbon, "Toward a 3-D camera based on single photon avalanche diodes", IEEE Journal of Selected Topics in Quantum Electronics, Vol. 10, No. 4, pp. 796-802, Jul./Aug. 2004.
- [5] A. Rochas, P.-A. Besse, and R. S. Popovic, "Actively recharged single photon counting avalanche photodiode integrated in an industrial CMOS process", Sensors and Actuators A: Physical, Vol.110, pp.124-129, Feb. 2004.
- [6] J. Hu, X. Xin, X. Li, J.H. Zhao, B.L. VanMil, K. Lew, R.L. Myers-Ward, C.R. Eddy, D.K. Gaskill, "4H-SiC Visible-Blind Single-Photon Avalanche

Diode for Ultraviolet Detection at 280 and 350 nm", IEEE Transactions on Electron Devices, Vol.55, No.8, pp.1977-1983, Aug. 2008.

- [7] M. Liu, C. Hu, J.C Campbell, Z. Pan, M.M Tashima, "Reduce Afterpulsing of Single Photon Avalanche Diodes Using Passive Quenching With Active Reset", IEEE Journal of Quantum Electronics, Vol. 44, No. 5, pp. 430-434, 2008.
- [8] A. Dalla Mora, A. Tosi, S. Tisa, F. Zappa, "Single-Photon Avalanche Diode Model for Circuit Simulations", IEEE Photonics Technology Letters, Vol.19, No.23, pp.1922-1924, Dec. 2007
- [9] P. Zhou, C. Liao, Z. Wei, C. Li and S. Yuan, "Determination of breakdown voltage of In 0.53Ga 0.47As/InP single photon avalanche diodes", Chinese Optics Letters, Vol.9, No.1, pp.0104021-0104023 Jan. 2011.
- [10] C. Hu, M. Liu, X. Zheng, and J.C. Campbell, "Dynamic Range of Passive Quenching Active Reset Circuit for Single Photon Avalanche Diodes", IEEE Journal of quantum electronics, Vol.46, No.1, pp.35-39, Jan. 2010.
- [11] D. Palubiak, M.M. El-Desouki, O. Marinov, M. J. Deen, and Q. Fang, "High-Speed, Single-Photon Avalanche-Photodiode Imager for Biomedical Applications", IEEE Sensors Journal, Vol.11, No.10, pp.2401-2412, Oct. 2011.
- [12] M. Mazzillo, G. Condorelli, D. Sanfilippo, G. Valvo, B. Carbone, G. Fallica, S. Billotta, M. Belluso, G. Bonanno, L. Cosentino, A. Pappalardo, and P. Finocchiaro, "Silicon Photomultiplier Technology at STMicroelectronics", IEEE Transactions on Nuclear Science, Vol.56 pp. 2434-2443, 2009.

- [13] A. Stewart, V. Saveliev, S. Bellis, D. Herbert, P. Hughes and J. Jackson, "Performance of 1-mm² Silicon Photomultiplier", IEEE Journal of Quantum Electronics, Vol. 44, pp.157-164, 2008.
- [14] E. Charbon, "Towards large scale CMOS single-photon detector arrays for lab-on-chip applications", Journal of Physics D: Applied Physics, Vol.41, No.9, 2008.
- [15] A. Nassalski, Member, M. Moszynski, A. Syntfeld-Kazuch, T. Szczeniak, S widerski, D. Wolski, T. Batsch and J. Baszak, "Multi Pixel Photon Counters (MPPC) as an Alternative to APD in PET Applications", IEEE Transactions on Nuclear Science, Vol.57, No.3, pp.1008-1014, Jun. 2010.
- [16] M. Entwistle, M.A. Itzler, J. Chen, M. Owens, K. Patel, X. Jiang, and S. Rangwala, "Geiger-mode APD camera system for single-photon 3D LADAR imaging", Proceeding of SPIE, Vol. 8375, 83750D-1, May. 2012.
- [17] H.T. Leem, K. H. Kim and Y. H. Chung, "CMOS APD array for multifunction of both sequential and summing mode operations", Journal of Instrumentation, Vol.6, C12053, Dec. 2011.
- [18] J. W. Mller, "Dead-time problems", Nuclear Instruments and Methods, Vol.112, No. 1-2, pp. 4757, 1973.
- [19] P.Antognetti, S.Cova and A.Longoni, "A study of the operation and performances of an avalanche diode as a single photon detector", Proc. 2nd Ispra Nuclear Electronics Symposium, pp.453-456, May 20-23, 1975.
- [20] S. Cova, M. Ghioni, A. Lacaita, C. Samori and F. Zappa, "Avalanche photodiodes and quenching circuits for single-photon detection", Applied optics, Vol.35, pp.1956-1976, 1996.

- [21] F. Zappa, M. Ghioni, S. Cova, C. Samori and A. Giudice, "An integrated active-quenching circuit for single-photon avalanche diodes", IEEE Transactions on Instrumentation and Measurement, Vol.49, pp.1167-1175, 2000.
- [22] F. Zappa, A. Lotito, A. Giudice, S. Cova and M. Ghioni, "Monolithic active-quenching and active-reset circuit for single-photon avalanche detectors", IEEE Journal of Solid-State Circuits, Vol.38, no.7, pp.1298-1301, 2003.
- [23] D. Cronin, A. M. Moloney, and A. P. Morrison. "Simulated Monolithically Integrated Single Photon Counter", IEEE High Frequency Postgraduate Student Colloquium, pp.9-14, Sep. 2004.
- [24] S. Tisa, F. Guerrieri and F. Zappa, "Variable-load quenching circuit for single-photon avalanche diodes", Optics Express, Vol.16, No.3, pp.2232-2244, Feb.2008.
- [25] R. Mita and G. Palumbo, "High-Speed and Compact Quenching Circuit for Single-Photon Avalanche Diodes", IEEE Transactions on Instrumentation and Measurement, Vol. 57, No. 3, pp.543-547, 2008.
- [26] Cristiano L. Niclass, "Single-Photon Image Sensors in CMOS: Picosecond Resolution for Three-Dimensional Imaging", Ph.D. thesis, Ecole polytechnique fdrale de Lausanne, 2008.
- [27] Mario Stipevi, "Active quenching circuit for single-photon detection with Geiger mode avalanche photodiodes", Applied Optics. Vol.48, pp.1705-1714, 2009
- [28] A. Sultana, E. Kamrani and M. Sawan, "CMOS silicon avalanche photodiodes for NIR light detection: a survey", Analog Integrated Circuits and Signal Processing, Vol.70, pp.1-13, 2012.

- [29] A. Rochas, P.-A. Besse, and R. S. Popovic, "Actively recharged single photon counting avalanche photodiode integrated in an industrial CMOS process", Sensors and Actuators A: Physical, Vol.110, pp.124-129, Feb. 2004.
- [30] A. Rochas, A. Pauchard, L. Monat, A. Matteo, P. Trinkler, R. Thew, and R. Ribordy, "Ultra-compact CMOS single photon detector, Advanced Photon Counting Techniques", 6372(1):63720N1-8, 2006.
- [31] C. Niclass and M. Soga, "A miniature actively recharged single-photon detector free of afterpulsing effects with 6ns dead time in a 0.18μm CMOS technology", 2010 IEEE International Electron Devices Meeting (IEDM), pp.14.3.1-14.3.4, Dec. 2010.
- [32] Andreas Eisele, Robert K. Henderson, Bernd Schmidtke, Tobias Funk, Lindsay Grant, Justin Richardson, Wolfgang Freude, "185 MHz Count Rate, 139 dB Dynamic Range Single-Photon Avalanche Diode with Active Quenching Circuit in 130nm CMOS Technology", 2011 International Image Sensor Workshop, pp.278-281, 2006.
- [33] D. Cronin and A. Morrison, "Intelligent System for Optimal Hold-Off Time Selection in an Active Quench and Reset IC", IEEE Journal of Selected Topics in Quantum Electronics, Vol.13, pp.911-918, 2007.
- [34] http://www.lfoundry.com/
- [35] Shijie Deng; A.P. Morrison, "High-resolution hold-off time control circuit for geiger-mode avalanche photodiodes", Proceedings of the 19th International Conference Mixed Design of Integrated Circuits and Systems (MIXDES), pp.269-272, May. 2012.
- [36] http://asic.austriamicrosystems.com/
- [37] http://www.europractice-ic.com/

- [38] D.M. Taylor, J.C. Jackson, A.P. Morrison, A. Mathewson and J.G. Rarity, "Characterization of novel active area silicon avalanche photodiodes operating in the geiger mode", Journal of Modern Optics, Vol.51, No.9/10, pp.1323-1332, Jul. 2004.
- [39] J.C. Jackson, J. Donnelly, B. ONeill, A.M. Kelleher; G. Healy, A.P. Morrison, A. Mathewson, "Integrated bulk/SOI APD sensor: bulk substrate inspection with Geiger-mode avalanche photodiodes", IET Electronics Letters, Vol.39, No.9, pp.735-736, May. 2003.
- [40] A.M. Moloney, A.P. Morrison, J.C. Jackson, A. Mathewson, J. Alderman, J. Donnelly, B. ONeill, A.M. Kelleher, G. Healy and P.J. Murphy, "Monolithically integrated avalanche photodiode and transimpedance amplifier in a hybrid bulk/SOI CMOS process", IET Electronics Letters, Vol.39, No.4, pp.391-392, Feb.2003.
- [41] S. Deng and A. P. Morrison, "Active quench and reset integrated circuit with novel hold-off time control logic for Geiger-mode avalanche photodiods", Optics Letters, Vol.37, No.18, pp.3876-3878, Sep 2012.
- [42] Shijie Deng and Alan P. Morrison. "Design of a Hold-off Time Control Circuit for Geiger-mode Avalanche Photodiodes", International Journal of Microelectronics and Computer Science, Vol.3, No.2, pp.53-59, 2012.

Chapter 5

Bias and gain control circuit for avalanche photodiodes

It was shown in Chapter 2 that the APD's multiplication gain is needs to be accurately controlled under varying environmental conditions for some APDbased applications. To accomplish this, a novel bias and gain control circuit is developed in this chapter for avalanche photodiodes (APDs) biased in linear mode. In the first section, some gain control and stabilization circuits for APDs are reviewed and the advantages and disadvantages for each circuit are discussed. Next, a bias and gain control circuit is designed and developed. This circuit is based on the idea of using two matched APDs to set and stabilize the gain which operates without the need for external temperature sensing and control electronics thereby lowering the system cost and complexity. With discrete components, a printed circuit board (PCB) module of the proposed circuit is capable of providing bias voltages up to 40 V for the APDs, precisely setting its gain and compensating for the changes in the temperature to maintain a stable gain.

5.1 Introduction to gain control circuit for avalanche photodiodes

In Chapter 2, it was seen that the APD gain is affected by many factors including reverse bias voltage and temperature. A number of reports have shown that APDs have significant temperature-dependent gain [1-5] which leads to a problem for the stability of APD-based systems. Therefore, it is necessary to precisely set the intended APD gain and maintain that gain for long periods of time under varying environmental conditions to ensure accurate performance in some APD-based applications such as those in nuclear medicine and astronomy [6-9].

Much research has focused on using temperature stabilization systems (thermal chambers or cooling systems) to provide a stable low APD temperature for APD gain stabilization as shown in [10-14]. However these solutions are usually very expensive in terms of initial and maintenance costs, thus limiting their usefulness to laboratory experiments. Figure 5.1 shows a more advanced temperature control system used for APD gain stabilization [15, 16]. This system uses a high frequency switching power regulator and a temperature control system to guarantee a low and extremely constant temperature in an APD. A temperature sensor in the thermal block is used to detect any variation of the temperature. The system receives feedback from the temperature sensor, compares it with an external reference signal and sends a control signal to the "Peltier cell" through the regulator to adjust the temperature of the APD. However, the use of the thermal block in this system makes it bulky, expensive and power intensive. Moreover, this system does not compensate for drift in the supply voltage that increases the inaccuracy of the APD gain stabilization.



Figure 5.1: Temperature control system in [15, 16].

A CPU-based gain stabilization system is proposed [8, 17] as shown in Figure 5.2. This design also uses a temperature sensor to detect the temperature variation in the APD. Unlike the system described in [15, 16], this system adjusts the APD's bias voltage instead of its temperature according to the feedback from the temperature sensor to keep the APD gain stable under moderate temperature variations. Results show that the circuit operates well over a temperature variation of $20^{\circ}C$. The active control of the APD gain by varying its bias as a function of the temperature has been used in a number of APD-based systems to operate the APDs independent of the temperature. Those designs can be seen in [7, 9, 18-20]. However, this method suffers from several drawbacks. The required bias voltages for the APD at different temperatures



Figure 5.2: Gain stabilisation system in [8, 17].

need to be measured in advance by the end user thus increasing the control complexity. Moreover, the temperature compensation elements suffer from temperature coefficient mismatches that further limit their operation accuracy.

An alternative control solution to set the APD gain is to use two matched APDs where one is biased at unitary gain (unmultiplied) and the other one is biased at a high gain (multiplied) [21, 22]; see Figure 5.3. Since the two matched APDs have near identical performance, the gain is simply the ratio of the currents flowing in the two APDs. In principle this technique can be used to set the gain and modify the bias voltage to compensate for the changes in the environment without the need for external temperature sensing and control electronics.



Figure 5.3: APD bias arrangement using two matched devices.

Based on this idea, many circuits have been designed. Figure 5.4 shows one of the APD gain control and stabilization circuits using this idea developed by the Photodetection and Imaging Group at University College Cork [22, 23].

In the circuit, APD1 is biased at unity gain which is independent of room temperature variations. APD2 is biased at high gain by an opto-coupler circuit which consists of an LED irradiating, an isolated feedback and an output PIN photodiode. A comparator is used to receive the feedback from the two matched APDs and send a signal to the FPGA for further processing. The FPGA receives the signal from the comparator and adjusts the high gain bias circuit through a digital-to-analogue converter to maintain V1 equal to V2 and thus maintain a stable multiplication gain in the APD2. This design operates without the need for external temperature sensing and control electronics. However, the control logic in this design is complicated and its bias solution



Figure 5.4: APD gain control and stablilization circuits in [22].

is not compact. As will be seen in the next section, a bias control and gain stabilization circuit is designed which has the advantages of using the matched APDs while providing a simpler and more compact solution.

5.2 Circuit description

Figure 5.5 shows the block diagram of the proposed circuit. A 5 V charge pump DC-DC converter is used to provide the high bias voltage up to 40 V. A gain control circuit is used with the bias circuit and the two matched APDs to set and stabilize the gain in the high gain APD.



Figure 5.5: Block diagram of proposed circuit.

5.2.1 Charge pump DC-DC converter

The DC-DC converter used is Dickson charge pump circuit [24], see Figure 5.6. In the circuit, an inverter is used to invert the phase of the input clocks. The neighbouring pumping capacitors are connected to these two inverse clocks. A load capacitor is used to smooth the output voltage which will be the only external component needed if the circuit is integrated on a chip. Through every stage, the positive charge is transferred by the diodes and pumping clocks to the output. At the output, a high voltage is obtained to bias the APD.



Figure 5.6: Charge pump DC-DC converter.

5.2.2 Gain control and stabilization

The schematic of the gain control circuit with two matched APDs is shown in Figure 5.7. APD2 is biased by the charge pump to generate a multiplied current, I_{MPh} . The other photodiode APD1 is biased to give an unmultiplied current, I_{Ph} .



Figure 5.7: Schematic of gain control circuit with two matched APDs.

In the gain control circuit, an op-amp and an NMOS-transistor are used to sense the voltage feedback from both photodiodes and control the bias voltage from the charge pump to adjust the gain of APD2. The bias voltages generate the unmultiplied and multiplied currents in APD1 and APD2 which flow through the load resistors R_{L1} and R_{L2} . Voltages V1 and V2 can be seen at the anode of both photodiodes. V1 is connected to the inverting input of an op-amp, Amp1 and V2 is connected to the non-inverting input of Amp1. If V2 exceeds V1, the op-amp generates an output voltage proportional to the difference between V2 and V1 and the NMOS-transistor M1 will be biased to divide the output current. The bias voltage of APD2 will be decreased and V2 will decrease. If V2 falls below V1, M1 will be turned off, the bias voltage of APD2 can be continuously charged up and V2 will increase. In this way, V2 can be set equal to V1 which makes the multiplication gain in APD2 equal to the ratio of R_{L1} and R_{L2} . When R_{L1} is fixed, the gain of APD2 can be altered by setting the value of the load resistor R_{L2} .

5.3 Simulations

For the simulations, an avalanche photodiode model shown in Figure. 5.8(a) is used. In the model, R1 and R2 represent the internal resistance when the APD is reversed biased far from and near the breakdown voltage which are set to 3 M Ω and 25 k Ω respectively. The voltage source V1 (set to 23 V) and diode D1 is used to model the large increase in gain when the APD is biased close to breakdown voltage. Figure 5.8(b) shows the simulation result of this model when the reverse bias is varied from 0 to 30V, demonstrating an approximate I-V characteristic for a planar APD with increasing gain approaching breakdown.



Figure 5.8: APD model and its simulation result.

The simulations of the proposed circuit were completed using the L-foundry 0.15 μ m CMOS process in the Cadence design environment [25]. An 8-stage charge pump is used with the pumping capacitors and load capacitor set to 20 pF and 1 nF, respectively. The clock frequency is 50 MHz with $V_{dd} = V_{clk} = 5$ V. APD1 is biased at V_{dd} to give unity gain and R_{L1} is set to 500 k Ω .



Figure 5.9: Gain and the bias voltage of APD2 when its gain is set to 100.

Figure 5.9 shows the gain and the bias voltage of APD2 when its gain is set to 100 ($R_{L2} = 5 \text{ k}\Omega$). As can be seen from the figure, the op-amp adjusts the bias voltage according to the feedback from the two photodiodes and control the gain precisely.

5.4 Development of PCB module and experimental results

With discrete components, a printed circuit board (PCB) was fabricated. Due to large parasitics and speed limitations of a discrete implementation, a larger load capacitor is used which is set to 47 nF to smooth the output voltage of the charge pump. The pumping capacitors are 22 pF and the diodes are Schottky diodes BAS40. The NMOS-transistor used is ZVN3306F and the op-amp is MCP6041. This circuit operates with clock frequency of 50 MHz and $V_{dd} = V_{clk} = 5$ V. A photograph of the fabricated PCB can be seen in Figure 5.10.



Figure 5.10: Photograph of the fabricated bias and gain control circuit gain.

Figure 5.11 illustrates the measured output voltages of the charge pump for different load currents. It shows the circuit is capable of providing up to 40 V bias voltage and delivering more than 1mA load current at a bias of 27 V.



Figure 5.11: Output voltages of charge pump DC-DC converter for different load currents.

For the measurement, two 400 μ m diameter circular planar shallow junction APDs developed by the Photodetection and Imaging Group from University College Cork were wire bonded in a dual in line (DIP) package; see Figure 5.12.



Figure 5.12: Packaged two matched avalanche photodiodes.

The block diagram of the experimental setup can be seen in Figure 5.13. A He-Ne laser source operating at 632.8 nm with the optical power of 330 nW was used as the incident light that was directed to the packaged diodes through an integrating sphere.



Figure 5.13: Block diagram of the experimental setup.

In the experiments, gain values of 20, 50 and 100 (R1/R2) were chosen to be the required gain values. Figure 5.14 shows the gain variation over a period of time. As can be seen from the figure, for the gains set to 20, 50 and 100, the mean gains measured are 20.55, 51.49 and 102.69 respectively. The gain offset errors are similar and close to 3% with standard deviations of less than 0.1. The gain offset error is mainly caused by the input bias current of the amplifier. Corrections can be made (Measured gain - 3% Set gain) to the measured gain so that it can demonstrate the intended set gain more accurately. Figure 5.15 shows the photo-current in the APD when the ambient temperature is changed. As shown in the figure, with the proposed circuit, the APD shows a more stable performance than that without the circuit. [26, 27].



Figure 5.14: Gain variation over a period of time under the control of the proposed circuit: (a) Gain set to 20; (b) Gain set to 50; (c) Gain set to 100.



Figure 5.15: APD's performance when ambient temperature is changed.

5.5 Conclusions

In this chapter, the development of a bias control and gain stabilization circuit has been described. Existing solutions in the literature were reviewed. The proposed circuit operation was explained; and simulations and measurements were presented. Temperature stabilization systems are used by many research groups to provide a stable low APD temperature for APD gain stabilization. These are expensive in terms of initial and maintenance costs and this limits their usefulness to a laboratory environment. Active control of the APD gain by varying its bias as a function of temperature is another way to operate the APDs with a stable gain, independent of the temperature. However, the required bias voltages for the APD at different temperatures must be measured in advance by the user. Moreover, the temperature compensation elements suffer from temperature coefficient mismatches that further limit their operational accuracy. The circuit developed in this work is capable of providing a high bias voltage for planar APDs, controlling and stabilizing their gain without the need for external temperature sensing and control elements thereby lowering the system cost and complexity. The circuit was simulated in the Cadence environment, fabricated with discrete components and tested using packaged APDs. Results show that the circuit can provide in excess of 40 V bias voltage for the planar APD and set the gain to within 3% of the desired gain with a standard deviation of less than 0.1.

Bibliography

- L.J.J.Tan, D.S.G. Ong, Jo Shien Ng, Chee Hing Tan, S.K. Jones, Yahong Qian and J.P.R. David, "Temperature Dependence of Avalanche Breakdown in InP and InAlAs", IEEE Journal of Quantum Electronics, Vol.46, No.8, pp.1153-1157, Aug. 2010.
- [2] Jo Shien Ng, Yu Ling Goh, Chee Hing Tan, Shiyong Zhang, J.P.R. David, "InAlAs Avalanche Photodiode With Type-II Superlattice Absorber for Detection Beyond 2um", IEEE Transactions on Electron Devices, Vol.58, No.2, pp.486-489, Feb. 2011.
- [3] Ian C. Sandall, Shiyu Xie, JingJing Xie and Chee Hing Tan, "High temperature and wavelength dependence of avalanche gain of AlAsSb avalanche photodiodes", Optics Letters, Vol.36, pp.4287-4289, 2011.
- [4] Chee Hing Tan, Shiyu Xie and Jingjing Xie, "Thin AlAsSb avalanche regions with sub-mV/K temperature coefficients of breakdown voltage and very low excess noise factors", IEEE Photonics Conference (PHO), pp.278-279, Oct. 2011.
- [5] Pin Jern Ker, P.R. David and Chee Hing Tan, "Temperature dependence of gain and excess noise in InAs electron avalanche photodiodes", Optics Express, Vol.20, No.28, pp.29568-29568, Dec.2012.
- [6] B.J. Pichler, F. Bernecker, G. Boning, M. Rafecas, W. Pimpl, M. Schwaiger, E. Lorenz and S.I. Ziegler, "A 48 APD array, consisting of two

monolithic silicon wafers, coupled to a 32-channel LSO matrix for high-resolution PET", IEEE Transactions on Nuclear Science, Vol.48, No.4, pp.1391-1396, 2001.

- [7] P.S. Marrocchesi, M.G. Bagliesi, K. Batkov, G. Bigongiari, M.Y. Kim, T. Lomtadze, P. Maestro, F. Morsani and R. Zei, "Active control of the gain of a Silicon PhotoMultiplier", Nuclear Instruments and Methods in Physics Research Section A: Accelerators, Spectrometers, Detectors and Associated Equipment, Vol.602, No.2, pp.391-395, Apr. 2009.
- [8] J. Kataoka, T. Toizumi, T. Nakamori, Y. Yatsu, Y. Tsubuku, Y. Kuramoto, T. Enomoto, R. Usui, N. Kawai, H. Ashida, K. Omagari, K. Fujihashi, S. Inagawa, Y. Miura, Y. Konda, N. Miyashita, S. Matsunaga, Y. Ishikawa, Y. Matsunaga and N. Kawabata, "In-orbit performance of avalanche photodiode as radiation detector on board the picosatellite Cute-1.7+APD", Journal of Geophysical Research, Vol.115, No.A5, pp., May 2010.
- [9] H. Anderhub, M. Backes, A. Biland, A. Boller, I. Braun, T. Bretz, S. Commichau, V. Commichau, D. Dorner, A. Gendotti, O. Grimm, H. von Gunten, D. Hildebrand, U. Horisberger, J.-H. Khne, T. Krhenbhl, D. Kranich, E. Lorenz, W. Lustermann, K. Mannheim, D. Neise, F. Pauss, D. Renker, W. Rhode, M. Rissi, M. Ribordy, U. Rser, L.S. Stark, J.-P. Stucki, O. Tibolla, G. Viertel, P. Vogler, Q. Weitzel, "A G-APD based Camera for Imaging Atmospheric Cherenkov Telescopes", Nuclear Instruments and Methods in Physics Research Section A: Accelerators, Spectrometers, Detectors and Associated Equipment, Vol.628, No.1, pp.107-110, Feb. 2011.
- [10] V.N. Solovov, V. Chepel, M.I. Lopes, R. Ferreira Marques, and A.J.P.L. Policarpo, "Study of large area avalanche photodiode for detecting liquid

Xenon scintillation", IEEE Transactions on Nuclear Science, Vol.47, No.4, pp.1307-1310, Aug. 2000.

- [11] M. Moszynski, W. Czarnacki, M. Szawlowski, B.L. Zhou, M. Kapusta, D. Wolski and P. Schotanus, "Performance of large-area avalanche photodiodes at liquid nitrogen temperature", IEEE Transactions on Nuclear Science, Vol.49, No.3, pp.971-976, Jun. 2002.
- [12] L. Yang, S.N. Dzhosyuk, J.M. Gabrielse, P.R. Huffman, C.E.H.Mattoni, S.E. Maxwell, D.N. McKinsey and J.M. Doyle, "Performance of a largearea avalanche photodiode at low temperature for scintillation detection", Nuclear Instruments and Methods in Physics Research A, pp.388-393, 2003.
- [13] A. Bieberle, E. Schleicher, U. Hampel, "Temperature control design for a high-resolution gamma-ray tomography detector", Review of Scientific Instruments, Vol.81, No.1, pp.014702-014707, Jan. 2010.
- [14] Makoto Akiba, Kenji Tsujino and Masahide Sasaki, "Ultrahigh-sensitivity single-photon detection with linear-mode silicon avalanche photodiode", Optics Letters, Vol.35, pp.2621-2623, 2010.
- [15] M.A. Perez, J.C. Campo, F.J. Ferrero and J.C.A. Anton, "Ultrastabilization of temperature in APD sensors by means a HF switching regulator", Proceedings of IEEE Sensors conference, pp.92-96, 2002.
- [16] M.A. Periz Garcia, J.C. Campo Rodriguez, N.A. Barragan Mendoza and J.C. Alvarez Anton, "Low-cost temperature stabilization in APD photo sensors by means a high frequency switching DC/T converter", IEEE Instrumentation and Measurement Technology Conference, pp.1733-1737, Aug. 2002.

- [17] J. Kataokaa, R. Satoa, T. Ikagawaa, J. Kotokua, Y. Kuramotoa, Y. Tsubukua, T. Saitoa, Y. Yatsua, N. Kawaia, Y. Ishikawab, and N. Kawabatab, "An active gain-control system for Avalanche photodiodes under moderate temperature variations", Nuclear Instruments and Methods in Physics Research A, 564(1), pp.300-307, 2006.
- [18] M.Y. Kim, C. Avanzini, M.G. Bagliesi, G. Bigongiari, T. Lomtadze, P. Maestro, P.S. Marrocchesi, F. Morsani and R. Zei, "Scintillation and Cherenkov light detection with a 3 mm 3 mm Silicon PhotoMultiplier", Nuclear Physics B Proceedings Supplements, Vol.197, No.1, pp.325-330, Dec. 2009.
- [19] Z. Kolka, P. Krivak and J. Kovar, "Adaptive voltage source for avalanche photodiodes", 19th International Conference Radioelektronika, pp.207-210, Apr. 2009.
- [20] Nan Zhang, J. Camp and M. Schmand, "Temperature compensation schemes for APD detectors in PET", IEEE Nuclear Science Symposium and Medical Imaging Conference (NSS/MIC), pp.2995-2996, Oct. 2011.
- [21] Y. Kuhara, M. Nishie and S. Hayashi, Optical receiver, US Patent, (2005,0082524), April 2005.
- [22] Diarmuid Patrick OConnell, "Control circuit for Avalanche Photodiodes", PhD thesis, University College Cork, Cork, Ireland, February 2009.
- [23] http://www.ucc.ie/en/eleceng/research/researchareas/photodectect/
- [24] J.F. Dickson, "On-chip high voltage generation in NMOS integrated circuits using an improved voltage multiplier technique", IEEE Journal of Solid State Circuits, Vol. SC-11, No. 3, pp. 374-378, 1976.
- [25] http://www.lfoundry.com/

- [26] S. Deng and A.P. Morrison, "A compact bias and gain control solution for avalanche photodiodes", IEEE Photonic Society 24th Annual Meeting (IPC11), pp. 282-283, Arlington, USA, October 2011.
- [27] Shijie Deng, John Hayes and Alan P. Morrison, "A Bias and Control Circuit for Gain Stabilization in Avalanche Photodiodes", The 23rd Irish Signals and Systems Conference (ISSC2012), Maynooth, Ireland, June 2012.

Chapter 6

Conclusions and future work

This chapter summarises the contributions achieved in this work and research ideas that could be pursued in the future are described.

6.1 Conclusions

In this thesis, three circuits have been designed and developed for control and enhancing the performance of the APD.

A bias solution that is used for the APDs operating in both linear and Geiger modes has been presented. The design consists of a dual-rail charge pump bias circuit and novel output voltage regulators. The operation of this circuit is evaluated experimentally and a layout of the circuit was designed using the L-Foundry 0.15 μ m process and post layout simulations are presented. Results show that the circuit is capable of sourcing mA range load currents for shallow-junction planar APDs that operate up to 40 V. The layout of the circuit designed using L-Foundry 0.15 μ m CMOS process shows a small footprint of 1.55 mm × 1 mm, making it suitable for hybrid integration with 2 external capacitors and an APD in a single package. This would operate seamlessly from a 5 V supply, thereby taking the additional responsibility for biasing the APD from the end-user. The possibility of developing a fully integrated version of this circuit was also discussed which shows that the circuit is capable of full integration for the devices with voltage ripple tolerance of greater than 70 mV.

A new active quench and reset circuit (AQRC) with novel hold-off time control logic for Geiger-mode avalanche photodiodes (GM-APDs) has been described. The circuit was designed and simulated in Cadence and sent for fabrication using a conventional 0.35 μ m CMOS process and tested with a GM-APD. Results show that this circuit is capable of linear changes to the hold-off time from several nanoseconds to microseconds with a reasonable step of 6.5 ns that allows setting of the optimal 'afterpulse-free' hold-off time for any GM-APD through digital inputs or additional signal processing circuitry. A minimum dead time of 28.4 ns was observed from the measurement, demonstrating a saturated photon-counting rate of 35.2 Mcounts/s with this AQRC.

A gain and stabilization control circuit has been demonstrated. This circuit is based on the idea of using two matched APDs to set and stabilize the gain which operates without the need for external temperature sensing and control electronics and thus lowers the system cost and complexity. A printed circuit board (PCB) module of the proposed circuit was fabricated with discrete components and measured with packaged APDs. Results show that the circuit is capable of providing bias voltages up to 40 V for the APDs, precisely setting its gain and compensating for the changes in the temperature to maintain a stable gain.

6.2 Future work

Following the developments from this thesis, some improvements could be made to the designed circuits and a number of areas could be pursued in the future research. Firstly, an on-chip clock generator could be designed and used in the bias circuit that will eliminate the need for the external clock signal. Filters can be applied at the output of the bias circuit that will decrease the ripples on the output bias voltage. The layout of the bias circuit should be submitted for fabrication. A hybrid package of the fabricated chip, an APD and several external components can provide a compact, single package solution (as shown in Figure 6.1) for APD-applications. A fully integrated bias circuit can be further developed for biasing the APD from a system power supply such as 5 V or 3.3 V.



Figure 6.1: Single package APD solution with bias circuit.

Secondly, there are potential improvements that can be made to the active quench and reset circuit. Integrating the active quench and reset circuit and the GM-APD would reduce the parasitics and the deadtime to achieve a higher saturated photon counting rate as discussed in Chapter 4. Newer CMOS processes can be used to design the circuit that will reduce the total chip area and give a faster speed response to the sensing circuit. Moreover, combining the bias circuit and active quench and reset circuit would produce a GM-APD system which is a highly compact solution with adjustable bias voltage and digitally controlled hold-off time. Signal processing circuitry could be added to the system for intelligent control of the bias voltage as well as the hold off time. In addition, this circuit could be also implemented with arrays of GM-APDs and some other circuitry to produce a complete imaging solution. When implementing the circuit with APD arrays, only one ring-oscillator is needed to provide the clocks which can be shared by the control logics for each pixel's hold-off time control that will simplify the design and reduce the circuit size.

Last but not least, the three circuits designed (bias circuit, active quench and reset circuit and gain stabilization circuit) can be combined. An interface logic needs to be designed. By controlling the bias voltage of the APD using the bias circuit, the APD's operation mode can be decided. When the APD is biased in linear mode then the gain control and stabilization circuit should be active to set the gain in the APD. When the APD is biased in Geiger-mode then the active quench and reset circuit should be active to enhance the GM-APD's performance. This system can be developed using existing circuits and implemented on a PCB module. It can also be designed into an integrated circuit that will decrease parasitics, improve speed response and reduce size, weight and power consumption. Read-out circuits and other signal processing circuitry also can be designed to process the results for each mode. Similarly, the circuit could be also implemented with APD arrays. The connections could be made in either hybrid or monolithic form.